



DISCRIMINATION, ARTIFICIAL INTELLIGENCE, AND ALGORITHMIC

Study by
Prof. Frederik Zuiderveen Borgesius
Professor of Law, Institute for Computing and Information
Sciences (iCIS), Radboud University Nijmegen, and Researcher
at the Institute for Information Law, University of Amsterdam
(the Netherlands)

The opinions expressed in this work are the responsibility of the author and do not necessarily reflect the official policy of the Council of Europe.

Published by the Directorate General of Democracy © Council of Europe - 2018 Printed in Strasbourg

TABLE OF CONTENTS

EXE	CUTIVE SUMMARY7
l.	INTRODUCTION9
II. DEC	ARTIFICIAL INTELLIGENCE AND ALGORITHMIC SISION-MAKING11
III.	DISCRIMINATION RISKS15
1.	HOW AI CAN LEAD TO DISCRIMINATION15
2.	FIELDS IN WHICH AI BRINGS DISCRIMINATION RISKS 23
IV.	LEGAL AND REGULATORY SAFEGUARDS31
1.	NON-DISCRIMINATION LAW32
2.	DATA PROTECTION LAW36
3.	OTHER REGULATION47
٧.	RECOMMENDATIONS50
1.	ORGANISATIONS USING AI50
2.	EQUALITY BODIES AND HUMAN RIGHTS MONITORING BODIES55
VI.	IMPROVING REGULATION61
1.	REGULATION AND FAST-DEVELOPING TECHNOLOGY61
2.	ENFORCEMENT64
3.	REGULATING NEW TYPES OF DIFFERENTIATION66
VII.	CONCLUSION
RIRI	IOGRAPHY 75

EXECUTIVE SUMMARY

This report, written for the Anti-discrimination department of the Council of Europe, concerns discrimination caused by algorithmic decision-making and other types of artificial intelligence (AI). AI advances important goals, such as efficiency, health and economic growth but it can also have discriminatory effects, for instance when AI systems learn from biased human decisions.

In the public and the private sector, organisations can take Aldriven decisions with far-reaching effects for people. Public sector bodies can use AI for predictive policing for example, or for making decisions on eligibility for pension payments, housing assistance or unemployment benefits. In the private sector, AI can be used to select job applicants, and banks can use AI to decide whether to grant individual consumers credit and set interest rates for them. Moreover, many small decisions, taken together, can have large effects. By way of illustration, AI-driven price discrimination could lead to certain groups in society consistently paying more.

The most relevant legal tools to mitigate the risks of Al-driven discrimination are non-discrimination law and data protection law. If effectively enforced, both these legal tools could help to fight illegal discrimination. Council of Europe member States, human rights monitoring bodies, such as the European Commission against Racism and Intolerance, and Equality Bodies should aim for better enforcement of current non-discrimination norms.

But AI also opens the way for new types of unfair differentiation (some might say discrimination) that escape current laws. Most non-discrimination statutes apply only to discrimination on the basis of protected characteristics, such as skin colour. Such statutes do not apply if an AI system invents new classes, which do not correlate with protected characteristics, to differentiate

between people. Such differentiation could still be unfair, however, for instance when it reinforces social inequality.

We probably need additional regulation to protect fairness and human rights in the area of Al. But regulating Al in general is not the right approach, as the use of Al systems is too varied for one set of rules. In different sectors, different values are at stake, and different problems arise. Therefore, sector-specific rules should be considered. More research and debate are needed.

I. INTRODUCTION

This report, written for the Anti-discrimination department of the Council of Europe, concerns risks of discrimination caused by algorithmic decision-making and other types of artificial intelligence (AI).

Al advances important goals, such as efficiency, health and economic growth. Our society relies on Al for many things, including spam filtering, traffic planning, logistics management, speech recognition, and diagnosing diseases. Al and algorithmic decision-making may appear to be rational, neutral and unbiased but, unfortunately, Al and algorithmic decision-making can also lead to unfair and illegal discrimination. As requested, the report focuses on the following questions.

- 1. In which fields do algorithmic decision-making and other types of AI create discriminatory effects, or could create them in the foreseeable future?
- 2. What regulatory safeguards (including redress mechanisms) regarding AI currently exist, and which safeguards are currently being considered?
- 3. What recommendations can be made about mitigating the risks of discriminatory AI, to organisations using AI, to Equality Bodies in Council of Europe member states, and to human rights monitoring bodies, such as the European Commission against Racism and Intolerance?
- 4. Which types of action (legal, regulatory, self-regulatory) can reduce risks?

This report uses the word "discrimination" to refer to objectionable or illegal discrimination, for instance on the basis of gender, skin colour, or racial origin. The report speaks of

.

¹ In line with legal tradition, I use the words "racial origin" and "race" in this report. However, I do not accept theories that claim that there are separate human races.

"differentiation" when referring to discrimination in a neutral, unobjectionable, sense.²

This report focuses on only one risk in relation to algorithmic decision-making and AI: the risk of discrimination. Many AI-related topics are thus outside the scope of this report, such as automated weapon systems, self-driving cars, filter bubbles, singularity, data-driven monopolies, the risk that AI or robots cause mass unemployment. Also out of scope are privacy-related questions regarding the massive amounts of personal data that are collected to power AI-systems.

The report relies on literature review. Because of length constraints, this report should be seen as a quick scan, rather than an in-depth mapping of all relevant aspects of Al, algorithmic decision-making, and discrimination. I would like to thank Bodó Balázs, Janneke Gerards, Dick Houtzager, Margot Kaminski, Dariusz Kloza, Gianclaudio Malgieri, Stefan Kulk, Linnet Taylor, Michael Veale, Sandra Wachter and Bendert Zevenbergen for their valuable suggestions.

The remainder of the report is structured as follows. Chapter II introduces artificial intelligence, algorithmic decision-making, and some other key phrases. Next, the report discusses the above-mentioned questions. Chapter III maps fields where AI leads or might lead to discrimination. Chapter IV discusses regulatory safeguards. Chapter V highlights how organisations can prevent discrimination when using AI. The chapter also offers recommendations to Equality Bodies and human rights monitoring bodies on mitigating the risks of discriminatory AI and algorithmic decision-making. Chapter VI gives suggestions on improving regulation, and chapter VII provides concluding thoughts.

_

² The purpose of algorithmic decision-making is often to discriminate (in the sense of differentiate or distinguish) between individuals or entities. See in detail the different meanings of "discrimination": Lippert-Rasmussen 2014.

II. ARTIFICIAL INTELLIGENCE AND ALGORITHMIC DECISION-MAKING

The phrases AI and algorithmic decision-making are used in various ways, and there is no consensus about definitions. Below artificial intelligence (AI), algorithmic decision-making and some related concepts are briefly introduced.

Algorithm

An algorithm can be described as "an abstract, formalised description of a computational procedure." ³ In this report, "decision" simply refers to the output, finding, or outcome of that procedure. As a rough rule of thumb, one could think of an algorithm as a computer program.

Sometimes, an algorithm decides in a fully automatic fashion. For instance, a spam filter for an e-mail service can filter out, fully automatically, spam messages from the user's inbox. Sometimes, humans make decisions assisted by algorithms; such decisions are *partly* automatic. For example, based on an assessment of a customer's credit by an AI system, a bank employee may decide whether a customer can borrow money from the bank.

However, when discussing discrimination, many risks are similar for fully and partly automated decisions. Recommendations by computers may have an air of rationality or infallibility, and people might blindly follow them. As Wagner et al. note, "the human being may often be led to "rubber stamp" an algorithmically prepared decision, not having the time, context or skills to make an adequate decision in the individual case." Human decision-makers may also try to minimise their own responsibility by following the computer's advice. The tendency to believe computers or to follow their advice is

³ Dourish 2016, p. 3. See also Domingos 2015.

⁴ Wagner et al. 2018, p. 8. See also Broeders et al. 2017, p. 24-25.

⁵ Zarsky 2018, p. 12.

sometimes called "automation bias".⁶ (We see in section IV.2 that some legal rules do distinguish fully and partly automated decisions.⁷)

Artificial intelligence

Artificial intelligence (AI) is, loosely speaking, "the science of making machines smart". 8 More formally, AI concerns "the study of the design of intelligent agents." 9 In this context, an agent is "something that acts", such as a computer. 10

Al is a broad research field, which exists since the 1940s. There are many types of Al. For instance, in the 1970s and 1980s, there was much research into "expert systems", "programs for reconstructing the expertise and reasoning capabilities of qualified specialists within limited domains. 12 Researchers programmed computers to answer questions, using preformulated answers. Such expert systems had some commercial success in the 1980s. 13 Expert systems had two disadvantages, observes Alpaydin. First, the logical rules in the systems did not always fit the messy reality of the world. "In real life, things are not true or false, but have grades of truth: a person is not either old or not old, but oldness increases gradually with age." 14 Second, experts had to provide the

⁶ Parasuraman and Manzey 2010. See also Citron 2007, p. 1271-1272; Rieke, Bogen and Robinson 2018, p. 11.

⁷ See the discussion of Article 22 GDPR in that section.

⁸ Royal Society 2017, p. 16.

⁹ Russel and Norvig 2016, p. 2, citing Poole, Mackworth and Goebel 1998, p. 1: "Computational Intelligence is the study of the design of intelligent agents."

¹⁰ Russel and Norvig 2016, p. 4.

¹¹ Two early publications are: Turing 1951 and McCarthy et al. 1955.

¹² Puppe 1993, p. 3.

¹³ Alpaydin 2016, p. 51.

¹⁴ Alpaydin 2016, p. 51.

knowledge (the answers) to put into the systems. That process costs a lot of time and money.¹⁵

Machine learning

In the past decade, one type of AI has been particularly successful: machine learning. ¹⁶ With machine learning, the knowledge in the system does not have to be provided by experts. "In contrast, machine learning systems are set a task and given a large amount of data to use as examples of how this task can be achieved or from which to detect patterns. The system then learns how best to achieve the desired output."¹⁷

As a rough rule of thumb, machine learning could be summarised as "data-driven predictions". ¹⁸ Lerh and Ohm give a more detailed description: "machine learning refers to an automated process of discovering correlations (sometimes alternatively referred to as relationships or patterns) between variables in a dataset, often to make predictions or estimates of some outcome. ¹⁹

Machine learning has become widely used during the past decade, in part because more and more data have become available to train the machines. Machine learning is so successful that nowadays many people say AI when they refer to machine learning (which is a type of AI).²⁰

Related phrases are data mining, big data and profiling. Data mining, a type of machine learning, is "the process of discovering interesting patterns from massive amounts of

¹⁶ Alpaydin 2016, p. 51. p. xiii.

¹⁵ Alpaydin 2016, p. 51.

¹⁷ Royal Society 2017, p. 19.

¹⁸ Paul, Jolley, and Anthony 2018, p. 6.

¹⁹ Lehr and Ohm 2017, p. 671. See also Royal Society 2017, p. 19.

²⁰ Lipton 2018; Jordan 2018.

data."²¹ Data mining is also referred to as "knowledge discovery from data".²² The phrase "big data" roughly refers to analysing large data sets. ²³ "Profiling" involves automated data processing to develop profiles that can be used to make decisions about people.²⁴

Terminology in this report

Regarding technology, this report sacrifices precision for readability, and uses "AI", "AI system", "AI decision" etc, without specifying whether AI refers to machine learning or another technology. Thus, in this report, an "AI system" can refer, for instance, to a computer running an algorithm that was fed data by its human operators.

For ease of reading, this report uses phrases such as "effects of AI", almost as if AI is an entity that acts on its own. However, AI systems do not spontaneously come into existence. As Wagner et al. note, "Mathematic or computational constructs do not by themselves have adverse human rights impacts but their implementation and application to human interaction does." Indeed, when an AI system makes decisions, it was an organisation that decided to use AI for that task.

In practice, an organisation that starts using AI rarely makes all relevant decisions about the AI system itself. An organisation might deploy an AI system, for which many important choices have been made already.²⁶ In some cases, the effects of certain decisions in a pre-procurement or design stage of an AI system

²¹ Han, Pei, and Kamber 2011, p. 33. See also Frawley et al. 1992, who describe data mining as "the nontrivial extraction of implicit, previously unknown, and potentially useful information from data."

²² Han, Pei, and Kamber 2011, p. xxiii. Some see data mining as one step in the "knowledge discovery" process.

²³ Boyd and Crawford 2012.

²⁴ See Hildebrandt 2008; Ferraris et al. 2013.

²⁵ Wagner et al 2018, p. 8. See also: Dommering 2006; Rieke, Bogen and Robinson 2018, p. 5.

²⁶ See, on the way that modern digital systems are developed: Gürses and Van Hoboken 2017. They focus on privacy, but their analysis is also relevant for AI systems and discrimination.

may only become apparent when the system is deployed in the real world. Apart from that, organisations can consist of many people, such as managers, lawyers and IT specialists. Nevertheless, for brevity, the report sometimes says that "organisations" do things. The next chapter discusses how AI can lead to discrimination and highlights areas where AI leads or might lead to discriminatory effects.

III. DISCRIMINATION RISKS

In which fields do algorithmic decision-making and other types of AI create discriminatory effects, or could create them in the foreseeable future?

1. HOW AI CAN LEAD TO DISCRIMINATION

This section discusses how AI can lead to discrimination; the next section gives examples where AI has led, or might lead, to discrimination. AI systems are often "black boxes".²⁷ It is often unclear for somebody why a system makes a certain decision about him or her. Because of the opaqueness of such decisions, it is difficult for people to assess whether they were discriminated against on the basis of, for instance, racial origin.

Al-driven decision-making can lead to discrimination in several ways. In a seminal paper, Barocas and Selbst distinguish five ways in which Al decision-making can lead, unintentionally, to discrimination. ²⁸ The problems relate to (i) how the "target variable" and the "class labels" are defined; (ii) labelling the training data; (iii) collecting the training data; (iv) feature selection; and (v) proxies. In addition, (vi), Al systems can be used, on purpose, for discriminatory ends. ²⁹ We discuss each problem in turn.

1) Defining the "target variable" and "class labels"

²⁷ Pasquale 2015.

²⁸ Barocas and Selbst 2016. See also O'Neil 2016, who gives an accessible and well-written introduction to discrimination and other risks in the area of Al systems.

²⁹ Barocas and Selbst 2016. They group the ways slightly differently.

Al involves computers that find correlations in data sets. For instance, when a company develops a spam filter, the company feeds the computer e-mail messages that are labelled by humans as "spam" and "non-spam". Those labelled messages are the training data. The computer finds which characteristics of e-mail messages correlate with being labelled as spam. The set of discovered correlations is often called "model" or "predictive model". For instance, messages that are labelled as spam might often contain certain phrases ("magic weight loss pill", "millions of dollars for you" etc), or might be sent from certain IP addresses. As Barocas and Selbst put it, "by exposing so-called "machine learning" algorithms to examples of the cases of interest (previously identified instances of fraud, spam, default, and poor health), the algorithm "learns" which related attributes or activities can serve as potential proxies for those qualities or outcomes of interest."30 Such an outcome of interest is called a "target variable".

"While the target variable defines what data miners are looking for", explain Barocas and Selbst, "'class labels' divide all possible values of the target variable into mutually exclusive categories." For spam filtering, people roughly agree about the class labels: which messages are spam or not. But for some situations, it is less obvious what the target variable should be. "Sometimes," note Barocas and Selbst, "defining the target variable involves the creation of *new* classes." Suppose a company wants an AI system to sort job applications to find good employees. How is a "good" employee to be defined? In other words: what should be the "class labels"? Is a good employee one who sells the most products? Or one who is never late at work?

³⁰ Barocas and Selbst 2016, p. 678.

³¹ Barocas and Selbst 2016, p. 678.

³² Barocas and Selbst 2016, p. 678-679, internal citations omitted.

³³ Barocas and Selbst 2016, p. 679.

Some target variables and class labels, explain Barocas and Selbst, "may have a greater or lesser adverse impact on protected classes." Suppose, for instance, that poorer people rarely live in the city centre and must travel further to their work than other employees. Therefore, poorer people are late for work more often than others because of traffic jams or problems with public transport. The company could choose "rarely being late often" as a class label to assess whether an employee is "good". But if people with an immigrant background are, on average, poorer and live further from their work, that choice of class label would put people with an immigrant background at a disadvantage, even if they outperform other employees in other aspects. In sum, discrimination can creep into an Al system because of how an organisation defines the target variables and class labels.

2) The training data: labelling examples

Al decision-making can also have discriminatory results if the system "learns" from discriminatory training data. Barocas and Selbst describe two ways in which biased training data can have discriminatory effects. First, the Al system might be trained on biased data. Second, problems may arise when the Al system learns from a biased sample.³⁶ In both cases, the Al system will reproduce that bias.

³⁴ Barocas and Selbst 2016, p. 680.

³⁵ See Peck 2013.

³⁶ Barocas and Selbst 2016, p. 680-681.

The training data can be biased because they represent discriminatory human decisions. Such a situation occurred at a medical school in the UK in the 1980s.³⁷ The school received many more applications than it could place. Therefore, the school developed a computer program to help sort the applications. The training data for the computer program were the admission files from earlier years, when people selected which applicants could enter medical school. The training data showed the computer program which characteristics (the input) correlated with the desired output (being admitted to the medical school). And the computer reproduced that selection system.

It turned out that the computer program discriminated against women and against people with an immigrant background. Apparently, in the years that provided the training data, the people that selected the students were biased against women and people with an immigrant background. As the British medical journal noted, "the program was not introducing new bias but merely reflecting that already in the system." In sum, if the training data are biased, the AI system risks reproducing that bias.

3) Training data: data collection

The sampling procedure can also be biased. For instance, when collecting data about crime, it could be the case that the police stopped more people with an immigrant background in the past. As Lum and Isaac note, "If police focus attention on certain ethnic groups and certain neighbourhoods, it is likely that police records will systematically over-represent those groups and neighbourhoods." 39

If an AI system is trained on such a biased sample, it will learn that people with an immigrant background are more likely to

³⁷ Lowry and Macpherson 1988; Barocas and Selbst 2016, p. 682.

³⁸ Lowry and Macpherson 1988.

³⁹ Lum and Isaac 2016, p. 15.

commit crime. Lum and Isaac note: "if biased data is used to train these predictive models, the models will reproduce (...) those same biases."⁴⁰

The effects of such a biased sample could even be amplified by AI predictions. Suppose the police pay extra attention in a neighbourhood with many immigrants, while that neighbourhood has average crime levels. The police register more crime in that neighbourhood than elsewhere. Because the numbers show more crime is registered (and thus seems to occur) in that neighbourhood, even more policemen are sent there. This way, policing on the basis of crime statistics can cause a feedback loop.⁴¹

To give another example: poor people may be underrepresented in a data set. This can be illustrated with Street Bump, a smartphone application that uses features such as GPS feeds to report road conditions to the city council. The Street Bump site explains: "Volunteers use the Street Bump mobile app to collect road condition data while they drive. The data provides governments with real-time information to fix problems and plan long-term investments."42 If there are fewer smartphone users among poor people than among wealthier people, poor people are likely to be undercounted. The effect could be that faulty roads in poor neighbourhoods are underrepresented in the dataset and therefore receive fewer reparations. The Street Bump app was used in the city of Boston, and that city aims to correct for such bias in data collection. 43 But the example illustrates how data collection could inadvertently lead to a biased data set. To sum up: biased training data can lead to biased AI systems.

⁴⁰ Lum and Isaac 2016, p. 15.

⁴¹ Lum and Isaac 2016, p. 16. See also Ferguson 2017; Harcourt 2008; Robinson and Koepke 2016.

⁴² http://www.streetbump.org accessed 10 September 2018.

⁴³ Crawford 2013. See also Barocas and Selbst 2016, p. 685; Federal Trade Commission 2016, p. 27.

4) Feature selection

A fourth problem relates to the features (categories of data) that an organisation selects for its AI system. If an organisation wants to use AI to predict something automatically, it needs to simplify the world to be able to capture it in data. ⁴⁴ As Barocas and Selbst note, an organisation must "make choices about what attributes they observe and subsequently fold into their analyses." ⁴⁵

Suppose that an organisation wants to predict automatically which job applicants will be good employees. It is not possible, or at least too costly, for an Al system to assess each job applicant completely. An organisation could focus, for instance, on certain features, or characteristics, of each job applicant.

By selecting certain features, the organisation might introduce bias against certain groups. For example, many employers in the US look for people who studied at famous and expensive universities. But it might be relatively rare for certain racial groups to study at those expensive universities. Therefore, it may have discriminatory effects if an employer selects job applicants on the basis of whether they studied at a famous university. ⁴⁶ In sum, organisations can cause discriminatory effects by selecting the features that an AI system uses for prediction.

⁴⁴ Barocas and Selbst 2016, p. 688.

⁴⁵ Barocas and Selbst 2016, p. 688.

⁴⁶ Barocas and Selbst 2016, p. 689.

5) Proxies

Another problem concerns proxies. Some data that are included in the training set may correlate with protected characteristics. As Barocas and Selbst point out, sometimes "criteria that are genuinely relevant in making rational and well-informed decisions also happen to serve as reliable proxies for class membership."⁴⁷

Suppose that a bank uses an AI system, trained on data covering the last twenty years, to predict which loan applicants will have problems repaying the loan. The training data do not contain information about protected characteristics such as skin colour. The AI system learns that people from postal code F-67075 were likely to default on their loans and uses that correlation to predict defaulting. Hence, the system uses what is at first glance a neutral criterion (postcode) to predict defaulting on loans. But suppose that the postcode correlates with racial origin. In that case, if the bank acted on the basis of this prediction and denied loans to the people in that postcode, the practice would harm people from a certain racial origin.

Barocas and Selbst explain that "[t]he problem stems from what researchers call "redundant encodings", cases in which membership in a protected class happens to be encoded in other data. This occurs when a particular piece of data or certain values for that piece of data are highly correlated with membership in specific protected classes."

To illustrate: a dataset that does not contain explicit data about people's sexual orientation can still give information about people's sexual orientation. "Facebook friendships expose sexual orientation", found a study from 2009. The study "demonstrates a method for accurately predicting the sexual orientation of Facebook users by analysing friendship

_

⁴⁷ Barocas and Selbst 2016, p. 691.

 $^{^{\}rm 48}$ Barocas and Selbst 2016, p. 692. See also Dwork et al 2012.

associations (...). [T]he percentage of a given user's friends who self-identify as gay male is strongly correlated with the sexual orientation of that user."⁴⁹

The proxy problem is difficult to solve. Barocas and Selbst note: "Computer scientists have been unsure how to deal with redundant encodings in datasets. Simply withholding these variables from the data mining exercise often removes criteria that hold demonstrable and justifiable relevance to the decision at hand." Hence, "[t]he only way to ensure that decisions do not systematically disadvantage members of protected classes is to reduce the overall accuracy of all determinations."

6) Intentional discrimination

Another situation can also occur: discrimination on purpose.⁵² For example, an organisation could intentionally use proxies to discriminate on the basis of racial origin. As Kroll et al. observe: "A prejudiced decisionmaker could skew the training data or pick proxies for protected classes with the intent of generating discriminatory results".⁵³ When an organisation uses proxies, the discrimination would be harder to detect than when the organisation openly discriminates.

To give a hypothetical example: an organisation could discriminate against pregnant women, while that discrimination would be difficult to discover. The US retail store Target reportedly constructed a "pregnancy prediction" score, based on around 25 products, by analysing the shopping behaviour of customers. If a woman buys some of those products, Target can predict with reasonable accuracy that she is pregnant. Target wanted to reach people with advertising during moments in life

⁴⁹ Jernigan and Mistree 2009.

⁵⁰ Barocas and Selbst 2016, p. 720.

⁵¹ Barocas and Selbst 2016, p. 721-722.

⁵² Barocas and Selbst 2016, p. 692. See also Bryson 2017; Friedman and Nissenbaum 1996; Hacker 2018, p. 1149; Kim 2017, p. 884; Vetzo, Gerards, and Nehmelman 2018, p. 145.

⁵³ Kroll et al. 2016, p. 682.

when they are more likely to change their shopping habits. Therefore, Target wanted to know when female customers were going to give birth. "We knew that if we could identify them in their second trimester, there's a good chance we could capture them for years". ⁵⁴ Target used the prediction for targeted marketing, but an organisation could also use such a prediction for discrimination. ⁵⁵

To sum up, AI decision-making can lead to discrimination in at least six ways, which relate to (i) the definition of the target variables and the class labels; (ii) the labelling and (iii) collecting of the training data; (iv) the selection of the features; (v) proxies. And (vi) organisations could use AI systems to discriminate on purpose. AI can also lead to other types of unfair differentiation, or to errors. We return to those topics in chapter VI.

2. FIELDS IN WHICH AI BRINGS DISCRIMINATION RISKS

This section provides examples of fields where AI decisionmaking has led, or could lead, to discrimination.

Police, crime prevention

We start with the public sector. A notorious example of an Al system with discriminatory effects is the system known as "Correctional Offender Management Profiling for Alternative Sanctions" – COMPAS for short. 56 The COMPAS system is used in parts of the US to predict whether defendants will commit crime again. The idea is that COMPAS can help judges to determine whether somebody should be allowed to go on probation (supervision outside prison). The COMPAS system does not use racial origin or skin colour as an input. But research by Angwin et al., investigative journalists at

 $^{^{54}}$ Duhigg 2012, quoting the statistician of Target. See on the Target case also Siegel 2013, Chapter 2.

⁵⁵ See Kim 2017, p. 884.

⁵⁶ See Equivant 2018.

ProPublica, showed in 2016 that COMPAS is "biased against blacks." FroPublica summarises:

COMPAS (...) correctly predicts recidivism 61 percent of the time. But blacks are almost twice as likely as whites to be labelled a higher risk but not actually reoffend. It makes the opposite mistake among whites: They are much more likely than blacks to be labelled lower risk but go on to commit other crimes ⁵⁸

Moreover, "Black defendants were also twice as likely as white defendants to be misclassified as being a higher risk of violent recidivism. And white violent recidivists were 63 percent more likely to have been misclassified as a low risk of violent recidivism, compared with black violent recidivists." 59

Northpointe, the company behind COMPAS, disputes that the system is unfair. ProPublica and Northpointe disagree mainly on what standard of fairness should be used to assess the system. Academic statisticians have argued that, in some cases, different standards of fairness are incompatible mathematically, which has consequences for what discrimination prevention should or could look like. ProPublica was concerned about what can be called "disparate mistreatment", where different groups receive different error types disproportionately (for instance individuals from some groups having a higher possibility of being deemed high-risk when they would not go on to commit a crime). Yet another

24

-

⁵⁷ Angwin et al 2016.

⁵⁸ Angwin et al 2016.

⁵⁹ Larson et al 2016.

⁶⁰ This paragraph is largely written by Michael Veale.

⁶¹ The discussion about COMPAS between ProPublica, Northpointe and academics is, in part, rather technical. A good summary of the discussion is: Feller et al. 2016. See also A shared statement of civil rights concerns 2018. See for the view of Northpointe: Dieterich, Mendoza and Brennan 2016.

important characteristic of risk scores is that they are correctly "calibrated". This means that for a group of individuals deemed to have an 80% chance of going on to commit a crime, 80% of that group indeed do go on to commit a crime. This should also be the same within groups, such as within black or white defendants. If this were not the case, then judges would need to interpret "high risk" for a black defendant differently than the same "high risk" for a white defendant, which brings other biases into play. Statisticians have indicated that where the underlying propensity to recidivism does differ, it is mathematically impossible to also have equalised error rates.⁶²

Sometimes the police use AI systems for predictive policing: automated predictions about who will commit crime, or when and where crime will occur. ⁶³ As noted above, predictive policing systems can reproduce and even amplify existing discrimination.

Selection of employees and students

In the private sector, AI can have discriminatory effects as well. We saw, for instance, that AI can be used to select prospective employees or students. As the example of the medical school in the UK showed, an AI system could lead to discrimination because of biased training data. Reportedly, Amazon stopped using an AI system for screening job applicants because the system was biased against women. In the words of Reuters, "the company realised its new system was not rating candidates for software developer jobs and other technical posts in a gender-neutral way." ⁶⁴ Based on historical training data, "Amazon's system taught itself that male candidates were preferable."

⁶² See Chouldechova 2017.

⁶³ Hildebrandt 2014; Ferguson 2017; Perry et al 2013; Van Brakel and De Hert 2011.

⁶⁴ Dastin 2018.

⁶⁵ Dastin 2018.

Advertising

Al is used for targeted online advertising, a very profitable sector for some companies (Facebook and Google, both among the world's most valuable companies, derive most of their profit from online advertising ⁶⁶). Online advertising can have discriminatory effects. Sweeney showed in 2013 that, when people searched for African-American-sounding names, Google displayed advertisements that suggested that somebody had an arrest record. For white-sounding names, Google displayed fewer ads suggestive of arrest records. Presumably, Google's Al system analysed people's surfing behaviour and inherited a racial bias.⁶⁷

Datta, Tschantz, and Datta simulated identical internet users who self-declared as male or female in settings. The researchers then analysed the ads that Google presented. Begin Google showed the simulated males ads from a certain career coaching agency that promised large salaries more frequently than the simulated females, a finding suggestive of discrimination. Begin Researchers note that it is unclear why women were shown fewer ads for high-paying jobs, because of the opaqueness of the system: We cannot determine who caused these findings due to our limited visibility into the ad ecosystem, which includes Google, advertisers, websites and users.

This is an example where the opaqueness of AI systems makes it harder to discover discrimination and its cause. People could be discriminated against without being aware. If an AI system

⁶⁶ Fortune 2018. The mother company of Google is officially called "Alphabet".

⁶⁷ Sweeney 2013.

⁶⁸ Datta, Tschantz and Datta 2015.

⁶⁹ Datta, Tschantz and Datta 2015, p. 93.

⁷⁰ Datta, Tschantz and Datta 2015, p. 92; Datta et al. 2018.

targets job ads only at men, women might not realise that they are excluded from the ad campaign.⁷¹

The Dutch Data Protection Authority found that Facebook enabled advertisers to target people based on sensitive characteristics. For instance, "data relating to sexual preferences were used to show targeted advertisements". 72 The Data Protection Authority says that Facebook amended its practices to make such targeting impossible. 73 Angwin and Perris, at ProPublica, showed that "Facebook lets advertisers exclude users by race. Facebook's system allows advertisers to exclude black, Hispanic and other "ethnic affinities" from seeing ads."74 ProPublica also showed that some firms use Facebook's targeting possibilities to advertise job ads only to people under a certain age. 75 Spanish researchers showed that "Facebook labels 73% of EU users with sensitive interests", such as "Islam", "reproductive health", and "homosexuality". 76 Advertisers can target advertising on the basis of such interests.

-

⁷¹ Munoz, Smith and Patil, 2016, p. 9; Zuiderveen Borgesius 2015a, chapter 3, section 3.

⁷² Dutch Data Protection Authority 2017; Dutch Data Protection Authority 2017a.

⁷³ Dutch Data Protection Authority 2017.

⁷⁴ Angwin and Perris 2016. See also Angwin, Tobin and Varner 2017. Dalenberg 2017 examines the application of EU non-discrimination law to ad targeting. In 2018, NGOs filed a lawsuit in the USA against Facebook for discrimination under US fair housing laws, for allowing the exclusion of women, disabled veterans and single mothers from a housing advertisement's potential audience (Bagli 2018).

⁷⁵ Angwin, Scheiber and Tobin 2017.

⁷⁶ Cabañas, Cuevas and Cuevas 2018. Such interests are defined as "special categories" of data, also called "sensitive data", in European data protection law. See article 9 of the European Union's General Data Protection Regulation. See, on data protection law: section IV.2.

Price discrimination

Online shops can differentiate the price for identical products based on information the shop has about a consumer: a practice called online price differentiation. A shop can recognise website visitors, for instance through cookies, and categorise them as price-sensitive or price-insensitive. With price differentiation, shops aim to charge each consumer the maximum price that he or she is willing to pay.⁷⁷

Princeton Review, a US company that offers online tutoring services, charged different prices in different areas in the US, ranging from 6600 to 8400 dollars. Presumably, the costs for delivering the service were the same for each area, as the company offers its tutoring service over the Internet. Angwin et al. found that the company's price differentiation practice led to higher prices for people with an Asian background: "Customers in areas with a high density of Asian residents were 1.8 times as likely to be offered higher prices, regardless of income." The company probably did not set out to discriminate on the basis of racial origin. Perhaps the company had tested different prices in different neighbourhoods and found that in certain areas people bought the same amount of services, even for higher prices. Nevertheless, the effect was that certain ethnic groups paid more.

Image search and analysis

Systems to search for images can also have discriminatory effects. In 2016, a search in Google Images for "three black teenagers" led to mugshots, while a search for "three white kids" mostly lead to pictures of happy white kids. In response to shocked reactions, Google said: "Our image search results are a reflection of content from across the web, including the frequency with which types of images appear and the way

⁷⁷ Zuiderveen Borgesius and Poort 2017.

⁷⁸ Angwin, Mattu and Larson 2015; Larson, Mattu and Angwin 2015.

they're described online. (...) This means that sometimes unpleasant portrayals of sensitive subject matter online can affect what image search results appear for a given query."⁷⁹ Indeed, one could say that Google's AI system merely reflected society.⁸⁰ But even if the fault lies with society rather than with the AI system, those image search results could influence people's beliefs.

Kay, Matuszek and Munson found that "image search results for occupations slightly exaggerate gender stereotypes and portray the minority gender for an occupation less professionally. There is also a slight under-representation of women."81

A different type of problem concerns image recognition by Al systems. Some image recognition software has difficulties in recognising and analysing non-white faces. Facial-tracking software by Hewlett Packard did not recognise dark-coloured faces as faces. Facial-tracking software by Hewlett Packard did not recognise dark-coloured faces as faces. Facial-tracking software by Hewlett Packard did not recognise dark-coloured faces as faces. Facial-tracking software by Hewlett Packard did not recognise dark-coloured faces as faces. Facial-tracking software by Hewlett Packard did not recognise dark-coloured faces as faces. Facial-tracking software by Hewlett Packard did not recognise dark-coloured faces. Facial-tracking software by Hewlett Packard did not recognise dark-coloured faces. Facial-tracking software by Hewlett Packard did not recognise dark-coloured faces. Facial-tracking software by Hewlett Packard did not recognise dark-coloured faces. Facial-tracking software by Hewlett Packard did not recognise dark-coloured faces. Facial-tracking software by Hewlett Packard did not recognise dark-coloured faces. Facial-tracking software by Hewlett Packard did not recognise dark-coloured faces. Facial-tracking software by Alikard Recognise dark-coloured faces. Facial-tracking software by Alikard Recognise dark-coloured faces. Facial-tracking software by Hewlett Packard did not recognise dark-coloured faces. Facial-tracking software by Hewlett Packard did not recognise dark-coloured faces. Facial-tracking software by Hewlett Packard did not recognise dark-coloured faces. Facial-tracking software by Hewlett Packard did not recognise dark-coloured faces. Facial-tracking software by Hewlett Packard did not recognise dark-coloured faces. Facial-tracking software by Hewlett Packard did not recognise dark-coloured faces as faces. Facial-tracking software by Hewlett Packard did not recognise dark-coloured faces as faces. Facial-tracking software faces as faces. Facial-tracking software faces as faces. Facial-tracking software

-

⁷⁹ Google's reaction, quoted in York 2016.

⁸⁰ Allen 2016.

⁸¹ Kay, Matuszek and Munson 2015.

⁸² Frucci 2009.

⁸³ BBC News 2015. See also Noble 2018.

⁸⁴ Sharp 2009.

⁸⁵ Regan 2016.

⁸⁶ Buolamwini and Gebru 2018.

Translation tools

The AI behind automated translation tools can also reflect inequality and discrimination. If people type "He is a doctor. She is a nurse" into Google Translate and translate the phrases into Turkish, Google Translate provides: "O bir hemşire. O bir doktor". Those Turkish sentences are gender-neutral; Turkish does not differentiate between the words "he" and "she". When translating the Turkish text into English again, Google Translate provides: "She is a nurse. He is a doctor".

The example is taken from research by Caliskan, Bryson and Narayanan, which shows "that machines can learn word associations from written texts and that these associations mirror those learned by humans." In other words, "natural language necessarily contains human biases, and the paradigm of training machine learning on language corpora means that Al will inevitably imbibe these biases as well."

Prates, Avelar and Lamb tested twelve gender-neutral languages, such as Hungarian and Chinese, in Google Translate. The authors wrote sentences such as "he/she is an engineer" in the gender-neutral languages and translated the sentences into English with Google Translate. The authors concluded that Google Translate "exhibits a strong tendency towards male defaults". ⁸⁹ Moreover, "male defaults are not only prominent but exaggerated in fields suggested to be troubled with gender stereotypes, such as STEM (Science, Technology, Engineering and Mathematics) jobs." ⁹⁰ In sum, Al-driven translation tools can provide results that reflect existing gender inequality. Perhaps such results could also worsen inequality, as they could influence people's ideas.

Nuancing the risks

⁸⁷ Caliskan, Bryson and Narayanan 2017.

⁸⁸ Narayanan 2016.

⁸⁹ Prates, Avelar and Lamb 2018, p. 1.

⁹⁰ Prates, Avelar and Lamb 2018, p. 28.

We saw that AI decision-making could have discriminatory effects – but AI systems do not necessarily perform worse than humans. Unfortunately, many humans also make discriminatory decisions. Indeed, in some cases, AI systems discriminate because they were trained on data that reflect discrimination by humans. Hence, it makes a difference whether one compares AI decision-making with human decisions in the real world (which, unfortunately, are sometimes discriminatory) or with hypothetical decisions in an ideal world without discrimination. Of course, the goal should be a world without any unfair or illegal discrimination.

Apart from that, AI could also be used to discover discrimination or inequality. Suppose an AI system shows that a collection of stock photos contains gender stereotypes. One way of interpreting such a finding is that the AI system illustrates stereotyped behaviour that already exists. Hence, an AI system could help to discover existing inequality that might have remained hidden otherwise.

IV. LEGAL AND REGULATORY SAFEGUARDS

What regulatory safeguards (including redress mechanisms) regarding AI currently exist, and which safeguards are currently being considered?

Non-discrimination law and data protection law are the main legal regimes that could protect people against Al-driven discrimination. This chapter discusses each regime in turn and highlights other potentially relevant fields of law and self-regulation. The chapter paints with a broad brush and focuses on the core principles of legal regimes. Issues lying outside the scope of this report include differences in regulation in Council of Europe member States, the territorial scope of laws and enforcement of laws against organisations in other States.

92 See Munoz, Smith and Patil 2016, p. 14.

⁹¹ See also Tene and Polonetsky 2017.

1. NON-DISCRIMINATION LAW

Discrimination is prohibited in many treaties and constitutions, including the European Convention on Human Rights. ⁹³ Article 14 of the European Convention on Human Rights states:

"The enjoyment of the rights and freedoms set forth in this Convention shall be secured without discrimination on any ground such as sex, race, colour, language, religion, political or other opinion, national or social origin, association with a national minority, property, birth or other status.⁹⁴"

Both *direct* and *indirect* discrimination are prohibited by the European Convention on Human Rights.⁹⁵ Direct discrimination means, roughly summarised, that people are discriminated against on the basis of a protected characteristic, such as racial origin. The European Court of Human Rights describes direct discrimination as follows: "there must be a difference in the treatment of persons in analogous, or relevantly similar,

-

⁹³ See e.g. Article 7 of the United Nations Declaration of Human Rights; Article 26 of the International Covenant on Civil and Political Rights; Article 21 of the Charter of Fundamental Rights of the European Union.

⁹⁴ Protocol 12 to that Convention lays down a similar prohibition, with, regarding certain aspects, a broader scope. "The enjoyment of any right set forth by law shall be secured without discrimination on any ground such as sex, race, colour, language, religion, political or other opinion, national or social origin, association with a national minority, property, birth or other status." Article 1, Protocol No. 12 to the Convention for the Protection of Human Rights and Fundamental Freedoms, European Treaty Series - No. 177, Rome, 4.XI.2000. On 18 September 2018, the total number of ratifications of/accessions to Protocol 12 stood at 20. See, for an up-to-date list: https://www.coe.int/en/web/conventions/search-on-treaties/conventions/treaty/177/signatures?p auth=0Kq9rtcm.

⁹⁵ While the European Convention on Human Rights has some horizontal effect, the Convention does not directly regulate discrimination in the private sector.

situations", which is based "on an identifiable characteristic". 96 EU law non-discrimination law uses a similar definition. 97

Indirect discrimination occurs, roughly speaking, when a practice is neutral at first glance but ends up discriminating against people of a certain racial origin (or another protected characteristic). ⁹⁸ Indirect discrimination is called "disparate impact" in the United States. Indirect discrimination is described as follows by the European Court of Human Rights:

"[A] difference in treatment may take the form of disproportionately prejudicial effects of a general policy or measure which, though couched in neutral terms, discriminates against a group. Such a situation may amount to "indirect discrimination", which does not necessarily require a discriminatory intent." 99

Indirect discrimination is defined similarly in EU law:

"Indirect discrimination shall be taken to occur where an apparently neutral provision, criterion or practice would put persons of a racial or ethnic origin at a particular disadvantage compared with other persons, unless that provision, criterion or practice is

⁹⁶ ECtHR, Biao v. Denmark (Grand Chamber), No. 38590/10, 24 May 2016, para. 89.

 $^{^{97}}$ Direct discrimination is defined as follows in Article 2(2)(a) of the Racial Equality Directive 2000/43/EC:

[&]quot;Direct discrimination shall be taken to occur where one person is treated less favourably than another is, has been or would be treated in a comparable situation on grounds of racial or ethnic origin." the Employment Equality Directive (2000/78/EC), the Gender Goods and Services Directive (2004/113/EC) and the Recast Gender Equality Directive (2006/54/EC) use similar definitions. But even within the European Union, non-discrimination law is only partly harmonised.

⁹⁸ See, generally on the concept of indirect discrimination: Tobler 2005; Ellis and Watson 2012, p. 148-155.

⁹⁹ ECtHR, Biao v. Denmark (Grand Chamber), No. 38590/10, 24 May 2016, para. 103.

objectively justified by a legitimate aim and the means of achieving that aim are appropriate and necessary."100

Al decision-making can unintentionally lead to indirect discrimination. Regarding indirect discrimination, the law focuses on the effects of a practice, rather than on the intention of the alleged discriminator. ¹⁰¹ Hence, it is not relevant whether the discriminator had the intention to discriminate.

Non-discrimination law can be used to fight discriminatory Al decisions. For instance, Al decisions that make people from a certain racial background pay more for goods and services could breach the prohibition of indirect discrimination. With Al decision-making, accidental indirect discrimination probably occurs more often than intentional discrimination.

However, non-discrimination law has several weaknesses in the context of Al decision-making. The prohibition of indirect discrimination does not provide a clear and easily applicable rule. The concept of indirect discrimination results in rather open-ended standards, which are often difficult to apply in practice. It needs to be proven that a seemingly neutral rule, practice or decision disproportionately affects a protected group and is thereby *prima facie* discriminatory. In many cases, statistical evidence is used to show such a disproportionate effect. 103

The European Court of Human Rights accepts that such a suspicion of indirect discrimination can be rebutted if the alleged discriminator can invoke an objective justification:

¹⁰⁰ Article 2(2)(b) of the Racial Equality Directive 2000/43/EC; capitalisation and punctuation adapted.

¹⁰¹ ECtHR, Biao v. Denmark (Grand Chamber), No. 38590/10, 24 May 2016, para. 103. See also Hacker 2018, p. 1153.

¹⁰² We could say: the prohibition of indirect discrimination is closer to a "standard" than to a "rule". See Sunstein 1995; Baldwin, Cave and Lodge 2011, chapter 14.

¹⁰³ ECtHR, D.H. and Others v. Czech Republic (Grand Chamber), No. 57325/00, 13 November 2007, paras. 187-188.

"A general policy or measure that has disproportionately prejudicial effects on a particular group mav be considered discriminatory where even it is not specifically aimed at that group and there is no discriminatory intent. This is only the case, however, if such policy or measure has "objective and reasonable" justification". 104

Such a justification must be objective and reasonable, and a measure, practice or rule does not meet these requirements if it:

"has no objective and reasonable justification, that is if it does not pursue a legitimate aim or if there is not a reasonable relationship of proportionality between the means employed and the aim sought to be achieved". 105

Along similar lines, EU law says that a practice will *not* constitute indirect discrimination if it "is objectively justified by a legitimate aim and the means of achieving that aim are appropriate and necessary". ¹⁰⁶ Whether an alleged discriminator can invoke such an objective justification depends on all the circumstances of a case and requires a nuanced proportionality test. ¹⁰⁷ Therefore, it is not always clear whether a certain practice breaches the prohibition of indirect discrimination.

¹⁰⁴ ECtHR, Biao v. Denmark (Grand Chamber), No. 38590/10, 24 May 2016, paras. 91 and 92.
I deleted internal citations and numbering from the quotation.

¹⁰⁵ ECtHR, Biao v. Denmark (Grand Chamber), No. 38590/10, 24 May 2016, para. 90. See also ECtHR, Case "relating to certain aspects of the laws on the use of languages in education in Belgium", No. 1474/62 and others, 23 July 1968, para. B.10.

¹⁰⁶ Article 2(2)(b) of the Racial Equality Directive 2000/43/EC.

¹⁰⁷ Collins and Khaitan 2018, p. 21; Hacker 2018, pp. 1161-1170.

The requirement that a *prima facie* case of indirect discrimination must be shown may also cause difficulties, since this type of discrimination can remain hidden. Suppose that somebody applies for a loan on the website of a bank. The bank uses an AI system to decide on such requests. If the bank automatically denies a loan to a customer on its website, the customer does not see why the loan was denied. Moreover, the customer cannot see whether the bank's AI system denies loans to a disproportionate percentage of, for instance, women. So even if customers knew that an AI system rather than a bank employee decided, it would be difficult for them to discover whether the AI system is discriminatory.

Another weakness relates to non-discrimination law's concept of protected characteristics. Non-discrimination statutes typically focus on (direct and indirect) discrimination based on protected characteristics, such as race, gender or sexual orientation. ¹⁰⁹ But many new types of Al-driven differentiation seem unfair and problematic – some might say discriminatory – while they remain outside the scope of most non-discrimination statutes. Hence, non-discrimination law leaves gaps. In section IV.3, we return to such unfair types of differentiation that might escape non-discrimination law.

In conclusion, non-discrimination law, in particular through the concept of indirect discrimination, prohibits many discriminatory effects of AI. However, enforcement is difficult, and non-discrimination law has weaknesses. The next section takes a look at data protection law.

2. DATA PROTECTION LAW

Data protection law is a legal tool that aims to defend fairness and fundamental rights, such as the right to privacy and the right

¹⁰⁸ See Larson et al 2017 for a similar example in real life: "These are the job ads you can't see on Facebook if you're older".

¹⁰⁹ Gerards 2007: Khaitan 2015.

to non-discrimination. ¹¹⁰ Data protection law grants rights to people whose data are being processed (data subjects) ¹¹¹ and imposes obligations on parties that process personal data (data controllers). ¹¹² Eight principles form the core of data protection law; they can be summarised as follows:

- (a) Personal data may only be processed lawfully, fairly and transparently ("lawfulness, fairness, and transparency").
- (b) Such data may only be collected for a purpose that is specified in advance, and should not be used for other unrelated purposes ("purpose limitation").
- (c) Such data should be limited to what is necessary for the processing purpose ("data minimisation").
- (d) Such data should be sufficiently accurate and up-to-date ("accuracy").
- (e) Such data should not be retained for an unreasonably long period ("storage limitation").
- (f) Such data should be secured against data breaches, illegal use etc ("integrity and confidentiality").¹¹³

¹¹⁰ See Article 1(2) and recital 71, 75, and 85 GDPR, and Article 1 of the COE Data Protection Convention 2018; Council of Europe Big Data Guidelines 2017, article 2.3.

¹¹¹ Article 4(1) GDPR; Article 2(a) COE Data Protection Convention 2018.

¹¹² Article 4(7) GDPR; Article 2(d) COE Data Protection Convention 2018.

¹¹³ Article 5(1)(a)-5(1)(f) GDPR; Articles 5, 7, and 10 COE Data Protection Convention 2018.

(g) The data controller is responsible for compliance ("accountability"). 114

These principles are included in the Council of Europe's Data Protection Convention 108 (revised in 2018 ¹¹⁵) and the European Union's General Data Protection regulation (GDPR, from 2016). Similar principles are included in more than a hundred national data privacy laws in the world. ¹¹⁶

Data protection law could help mitigate risks of unfair and illegal discrimination. ¹¹⁷ For instance, data protection law requires transparency about personal data processing. Therefore, organisations must provide information, for instance in a privacy notice, about all stages of an Al decision-making process that involve personal data. ¹¹⁸ It is true that most people do not read privacy notices. ¹¹⁹ Nevertheless, such notices could be helpful for researchers, journalists, and supervisory authorities. If a privacy notice suggests that a processing practice could have discriminatory effects, authorities can investigate.

Under certain circumstances, the GDPR and Data Protection Convention 108 require organisations (data controllers) to conduct a data protection impact assessment (DPIA). An impact assessment can be described as follows:

An impact assessment is a tool used for the analysis of possible consequences of an initiative on a relevant societal concern or concerns, if this initiative can present dangers to these concerns, with a view to

¹¹⁴ Article 5(2) of the GDPR; Article 10(1) COE Data Protection Convention 2018.

¹¹⁵ Article 5, 7, and 10 COE Data Protection Convention 2018.

¹¹⁶ Greenleaf 2017.

¹¹⁷ See, on the interplay between data protection law and discrimination law: Schreurs et al. 2008; Gellert et al. 2013; Hacker 2018; Lammerant, De Hert, Blok 2017.

¹¹⁸ Article 5(1)(a); Article 13; Article 14 GDPR; Articles 5(4)(a) and 8 COE Data Protection Convention 2018.

¹¹⁹ Zuiderveen Borgesius 2015.

supporting informed decision-making whether to deploy this initiative and under what conditions, ultimately constituting a means to protect these concerns.¹²⁰

The GDPR requires a DPIA when a practice is "likely to result in a high risk to the rights and freedoms of natural persons", especially when using new technologies. ¹²¹ In some circumstances, the GDPR always requires a DPIA (because the GDPR assumes a high risk), for instance when organisations take fully automated decisions that have legal or similar effects for people. ¹²² Hence, for many AI systems that make decisions about people, the GDPR requires a DPIA. ¹²³ The risk of unfair or illegal discrimination must also be considered when conducting a DPIA. ¹²⁴

Under the Council of Europe's Data Protection Convention 108, and under the Charter of Fundamental Rights of the European Union, each member State must have an independent Data Protection Authority. Such Data Protection Authorities must have powers of investigation. The GDPR gives most details about the investigative powers of Data Protection Authorities. A Data Protection Authority can, for instance, obtain access to premises of controllers, carry out investigations in the form of data protection audits and order data controllers to provide

¹²⁰ Kloza et al. 2017, p. 1. See also Article 29 Working Party 2017 (WP248); Binns 2017; Mantelero 2017; Wright and De Hert 2012.

¹²¹ Article 25(1) GDPR.

¹²² Article 35(3)(a) GDPR. See also recital 91 GDPR.

¹²³ Article 35(3)(b) and 35 (3)(c) GDPR could also apply to some AI systems.

¹²⁴ Article 29 Working Party 2017 (WP248), p. 6, p. 14. See also Kaminski 2018a, p. 25; Edwards and Veale 2017.

¹²⁵ Article 8(3) of the Charter of Fundamental Rights of the European Union. See also Article51 GDPR; chapter IV COE Data Protection Convention.

¹²⁶ Chapter VI GDPR; chapter IV COE Data Protection Convention.

information and to give access to their data processing systems. 127

Rules on automated decisions

The GDPR contains specific rules for certain types of "automated individual decision-making". ¹²⁸ These rules aim, among other things, to mitigate the risk of illegal discrimination. ¹²⁹ The Council of Europe's Data Protection Convention also contains rules on automated decisions, which are less detailed than in the GDPR. ¹³⁰ Here, we focus on the GDPR.

Article 22 of the GDPR, sometimes called the Kafka provision, contains an in-principle prohibition of fully automated decisions with legal or similar significant effects and applies, for instance, to fully automated e-recruiting practices without human intervention. The predecessor of the GDPR already had a similar provision, which has not been applied much in practice. The main rule of the GDPR's provision on automated individual decision-making reads as follows:

The data subject shall have the right not to be subject to a decision based solely on automated processing, including profiling, ¹³³

¹²⁷ Article 58(1) GDPR. The Data Protection Authority can also exercise these rights against "processors", organisations that process personal data for data controllers.

¹²⁸ Article 22 GDPR. The discussion of the GDPR's rules on automated decisions is based on and includes sentences from Zuiderveen Borgesius and Poort 2017.

¹²⁹ See Recital 71 GDPR.

¹³⁰ Article 9(1)(a) COE Data Protection Convention 2018.

¹³¹ Recital 71 GDPR.

¹³² Korff 2012. The predecessor was Article 15 of the Data Protection Directive. That Article 15 was based on a provision of the Data Protection Act of France from 1978. See Bygrave 2001.

¹³³ The GDPR defines "profiling" as follows: "'Profiling' means any form of automated processing of personal data consisting of the use of personal data to evaluate certain personal aspects relating to a natural person, in particular to analyse or predict aspects concerning that natural person's performance at work, economic situation, health, personal preferences, interests, reliability, behaviour, location or movements." Art. 4(4) GDPR.

which produces legal effects concerning him or her or similarly significantly affects him or her ¹³⁴

Roughly summarised: people may not be subjected to certain automated decisions with far-reaching effects. The GDPR says people have a "right not to be subject to" certain decisions. But it is generally assumed that this right implies an in-principle prohibition of such decisions.¹³⁵

Slightly rephrasing Mendoza and Bygrave, four conditions must be met for the provision to apply: (i) there is a decision, which is based (ii) solely (iii) on automated data processing; (iv) the decision has legal or similarly significant effects for the person.¹³⁶

An example of a decision with "legal effects" would be a court decision, or decision regarding a social benefit granted by law, such as pension payments. An example of a decision with "similarly significantly" effects would be a bank that denies credit automatically. And Data Protection Authorities say that online price differentiation could "similarly significantly affect" somebody, if it leads to "prohibitively high prices [that] effectively bar someone from certain goods or services.

There are exceptions to the in-principle prohibition of certain automated decisions. In short, the prohibition does not apply if the automated decision (i) is based on the individual's explicit

¹³⁵ De Hert and Gutwirth 2008; Korff 2012; Wachter, Mittelstadt, and Floridi 2017; Zuiderveen Borgesius 2015a.

¹³⁷ See Article 29 Working Party 2018 (WP251), p. 21.

¹³⁴ Art. 22 GDPR.

¹³⁶ Mendoza and Bygrave 2017.

¹³⁸ Recital 71 GDPR. See for more examples that "could" constitute automated decisions that "similarly significantly affect" people: Article 29 Working Party 2018 (WP251), p. 22.

¹³⁹ Article 29 Working Party 2018 (WP251). p. 22.

consent; (ii) is necessary for a contract between the individual and the data controller; or (iii) is authorised by law. 140

If a controller can rely on the (i) consent or (ii) contract exception to bypass the prohibition, a different rule is triggered: "the data controller shall implement suitable measures to safeguard the data subject's rights and freedoms and legitimate interests, at least the right to obtain human intervention on the part of the controller, to express his or her point of view and to contest the decision". ¹⁴¹ Hence, in some circumstances, the data subject can ask for a human to reconsider the automated decision. For instance, a bank could ensure that customers can call the bank to have a human reconsider the decision, if the bank automatically denies them a loan through the bank's website.

In addition to its general transparency requirements, the GDPR also contains transparency requirements specific to automated decisions:

[T]he controller shall provide the data subject with the following information (...) the existence of automated decision-making, including profiling (...) and, at least in those cases, meaningful information about the logic involved, as well as the significance and the envisaged consequences of such processing for the data subject.¹⁴²

Hence, in some cases, an organisation would have to explain that it uses AI decision-making and would have to provide meaningful information about the logic of that process.

There has been a great deal of scholarly attention as to whether the GDPR's rules on automated decisions create a "right to

¹⁴⁰ Article 29 Working Party 2018 (WP251), p. 22.

¹⁴¹ Article 22(3) GDPR. As Kaminski notes, the GDPR's text "creates a version of algorithmic due process: a right to an opportunity to be heard." Kaminski 2018a, p. 8.

¹⁴² Article 13(2)(f) and 14(2)(f) GDPR.

explanation" of individual decisions. Recital 71 suggests the existence of an individual right to "explanation" of Al decisions – a right that could be useful to protect fairness. 44

Many scholars are sceptical of whether such a right would be effective, noting for instance that many types of automated decisions remain outside the scope of the GDPR's rules. 145 To illustrate: the GDPR's automated decision provision only applies to decisions based "solely" on automated processing. Hence, when a bank employee denies a loan on the basis of a recommendation by an Al system, as long as the employee is not rubber-stamping, the provision does not apply. 146

¹⁴³ See for instance Edwards and Veale 2017; Goodman and Flaxman 2016; Kaminski 2018; Kaminski 2018a; Malgieri G and Comandé 2017; Mendoza and Bygrave 2017; Selbst and Powles 2017; Wachter et al. 2017.

¹⁴⁴ Recital 71: "such processing should be subject to suitable safeguards, which should include (..) the right (...) to obtain an explanation of the decision reached after such assessment".

¹⁴⁵ See for instance Edwards and Veale 2017; Wachter et al. 2017; Zuiderveen Borgesius 2015a, chapter 9, section 6.

¹⁴⁶ See Article 29 Working Party 2018 (WP251). The Working Party says that a superficial check by a human (rubber stamping) is not sufficient. However, as noted by Veale and Edwards 2018, it is not clear how organisations are supposed to ensure that decisions have non-superficial human input.

It remains to be seen what the practical effect of these GDPR provisions will be. As noted, the predecessor of the GDPR provision on automated decisions has remained a dead letter. Regardless, the attention to the GDPR provisions has helped to foster an interdisciplinary discussion on explaining AI decisions.

The modernised Convention 108 appears more generous for individuals in its phrasing around explanation rights. Unlike the GDPR provision, which applies to decisions that have significant effect and are "solely" based on automated processing, Convention 108 gives individuals a right "to obtain, on request, knowledge of the reasoning underlying data processing where the results of such processing are applied to him or her". The breadth of what it means to "apply" a "result" is yet to be seen in practice in any national implementations.

Caveats

Several caveats are in order regarding data protection law's possibilities as a tool to fight Al-driven discrimination. First, there is a compliance and enforcement deficit. Data Protection Authorities have limited resources. And many Data Protection Authorities do not have the power to impose serious sanctions (in the EU, such authorities received new powers with the GDPR). Previously, many organisations did not take compliance with data protection law seriously. ¹⁴⁸ It appears that compliance improved with the arrival of the GDPR, but it is too early to tell.

Second, parts of algorithmic processes are outside the scope of data protection law. Data protection law only applies when personal data are processed. It does not apply to predictive models because they do not relate to identifiable persons. For example, a predictive model that says "80% of the people living

¹⁴⁷ Article 9(1)(c) COE Data Protection Convention 2018: "Every individual shall have a right (...) to obtain, on request, knowledge of the reasoning underlying data processing where the results of such processing are applied to him or her". See Veale and Edwards 2018.

¹⁴⁸ See Zuiderveen Borgesius 2015a, chapter 8, section 2.

in postal code F-67075 pay their bills late" is not a personal datum, as the model does not refer to an individual. (When a predictive model is applied to an individual, data protection law applies again. 149)

Third, data protection law uses many open and abstract norms, rather than black-and-white rules.¹⁵⁰ Data protection law must use open norms, because its provisions apply in many different situations, in the private and the public sector. This regulatory approach, an omnibus approach, has many advantages. For instance, the open norms do not have to be adapted each time when a new technology is developed. But one disadvantage is that the open norms can be difficult to apply.¹⁵¹

Fourth, data protection law has strict rules on "special categories" of data (sometimes called "sensitive data"), such as data regarding racial origin or revealing health status. Those rules create challenges for assessing and mitigating discrimination. Many of the methods to tackle discrimination in AI systems implicitly assume that organisations hold these sensitive data — yet to meet data protection law, many organisations may not be holding them. Tension remains between respecting data protection law and collecting sensitive data to fight discrimination.

Fifth, even where explanations of AI decisions might be legally required by the GDPR or Convention 108, it is often difficult to explain the logic behind a decision, when an AI system,

¹⁴⁹ See Zuiderveen Borgesius 2015a, chapter 2 and chapter 5. See, on the weaknesses of data protection law in the area of Al decision-making: Wachter and Mittelstadt 2018.

¹⁵⁰ Zuiderveen Borgesius 2015a, chapter 9, section 1.

¹⁵¹ See, on different types of legal rules: Chapter VI, section 1.

¹⁵² Article 9 GDPR; article 6 COE Data Protection Convention 2018. The strict rules for special categories of data aim, in part, to fight discrimination. See on "special categories of data" in the context of Al: Malgieri and Comandé 2017a.

¹⁵³ Goodman 2016; Ringelheim and De Schutter 2008; Ringelheim and De Schutter 2009; Veale and Binns 2017; Žliobaitė and Custers 2016. Some methods to audit Al systems while maintaining privacy using cryptography are emerging. See Kilbertus et al. 2018.

analysing large amounts of data, arrives at that decision.¹⁵⁴ And in some cases, it is not clear how much an explanation would help people, especially insofar as it places the burden on them to understand the decision and its appropriateness.¹⁵⁵

That said, more transparency and explanation of AI decisions could be useful. For more than a decade, scholars have been calling for the development of transparency-enhancing technologies (TETs), to enable meaningful transparency regarding automated decision-making. ¹⁵⁶ Such technologies should "aim at making information flows more transparent through feedback and awareness thus enabling individuals as well as collectives to better understand how information is collected, aggregated, analysed and used for decision-making." ¹⁵⁷ Computer scientists are exploring various forms of explainable AI. ¹⁵⁸

In any case, it is much too early to assess the effect of the modernised Convention 108 and the GDPR. More legal research is needed on how data protection law could help to mitigate discrimination risks. ¹⁵⁹ While data protection law is largely untested as a non-discrimination tool, it does offer possibilities to fight illegal discrimination.

-

¹⁵⁴ Ananny and Crawford 2016; Burrell 2016; Binns et al. 2018; Edwards and Veale 2017; Hildebrand 2015; Kroll et al. 2016; 2018; Wachter, Mittelstadt and Russell 2017.

¹⁵⁵ Edwards and Veale 2017.

¹⁵⁶ Hildebrandt and Gutwirth 2008, chapter 17.

¹⁵⁷ Diaz and Gürses 2012.

¹⁵⁸ See Guidotti et al. 2018; Miller 2017; Selbst and Barocas 2018; Tickle et al. 1998. See also this Google project: "What If... you could inspect a machine learning model, with no coding required?", https://pair-code.github.io/what-if-tool/index.html#about accessed 1 October 2018. That project took inspiration from Wachter, Mittelstadt and Russell 2017.

¹⁵⁹ Researchers are starting to explore how data protection law can help to fight discrimination. See for instance: Goodman 2016; Mantelero 2018; Hacker 2018; Hoboken and Kostic (forthcoming); Wachter 2018; Wachter and Mittelstadt 2018.

3. OTHER REGULATION

In the area of AI decisions, other fields of law could also help to ensure fairness, and perhaps help to mitigate discrimination-related problems. For example, consumer law could be invoked to protect consumers against some types of manipulative AI-driven advertising. ¹⁶⁰ As discriminatory behaviour by a company causes more problems when the company has a monopoly position, competition law could also help to protect people. ¹⁶¹ For the public sector, administrative law and criminal law could be relevant to protect fair procedures. ¹⁶² Freedom of information laws could be used to obtain information about public sector AI systems. ¹⁶³ But the application of these fields of law to protect people in the area of AI is largely unexplored. A discussion of those fields of law falls outside the scope of this report.

Regulation under consideration

Several regulatory measures that could be relevant for Al-driven discrimination are currently being considered. The Council of Europe's Consultative Committee of the Convention for the Protection of Individuals with regard to Automatic Processing of Personal Data published a draft report in September 2018: "Artificial intelligence and data protection: challenges and possible remedies."

The Council of Europe's Steering Committee on Media and Information Society has set up an expert committee on AI: the Committee of Experts on human rights dimensions of automated data processing and different forms of artificial

47

¹⁶⁰ See, on Al and consumer law: European Data Protection Supervisor 2014; Helberger, Zuiderveen Borgesius and Reyna 2017; Jablonowska et al. 2018.

¹⁶¹ See, on Al and competition law: Ezrachi and Stucke 2016; Graef 2016; Graef 2017; Valcke, Graef and Clifford 2018; Van Nooren et al. 2018.

¹⁶² See, on Al and administrative law: Van Eck 2018; Oswald 2018, Cobbe 2018.

¹⁶³ See Rieke, Bogen and Robinson 2018, p. 24; Fink 2018; Oswald and Grace 2016.

¹⁶⁴ Mantelero 2018.

intelligence. The expert committee will conduct studies and give guidance for possible future standard-setting.¹⁶⁵

The European Union is active in the area of AI too. In 2018, the European Commission published a communication on AI, and has set up a High-Level Expert Group on Artificial Intelligence, ¹⁶⁶ which is tasked with proposing draft AI Ethics Guidelines. ¹⁶⁷ The EU Agency for Fundamental Rights is also examining AI. ¹⁶⁸ Furthermore, in 2017 the Commission proposed an ePrivacy Regulation to protect privacy on the Internet, which could be relevant for AI and machine learning, as it would limit the collection of certain types of privacy-sensitive data on the Internet. ¹⁶⁹

An EU Regulation from 2016 concerns one type of AI decision: algorithmic trading on stock exchanges etc. The Regulation states: "An investment firm shall ensure that its compliance staff has at least a general understanding of how the algorithmic trading systems and trading algorithms of the investment firm operate." Moreover, "an investment firm shall establish and monitor its trading systems and trading algorithms through a clear and formalised governance arrangement". The Perhaps similar requirements could be adopted for other sectors.

Self-regulation

^{. .}

¹⁶⁵ Council of Europe MSI-AUT 2018.

https://ec.europa.eu/digital-single-market/en/high-level-expert-group-artificial-intelligence accessed 26 September 2018.

¹⁶⁷ European Commission, Artificial Intelligence for Europe, 2018, p. 16. See also: European Group on Ethics in Science and New Technologies 2018.

http://fra.europa.eu/en/project/2018/artificial-intelligence-big-data-and-fundamental-rights accessed 13 October 2018.

¹⁶⁹ See Zuiderveen Borgesius et al 2017.

¹⁷⁰ Article 2, Commission Delegated Regulation (EU) 2017/589 of 19 July 2016 supplementing Directive 2014/65/EU of the European Parliament and of the Council with regard to regulatory technical standards specifying the organisational requirements of investment firms engaged in algorithmic trading.

¹⁷¹ Article 1. idem.

Several organisations have proposed principles that aim for fair, accountable or ethical AI. For example, the organisation FATML, Fairness, Accountability, and Transparency in Machine Learning, published "Principles for accountable algorithms and a social impact statement for algorithms" The principles call for organisations to "ensure that algorithmic decisions do not create discriminatory or unjust impacts when comparing across different demographics (eg race, sex, etc)." The principles call for organisations to "ensure that algorithmic decisions do not create discriminatory or unjust impacts when comparing across different demographics (eg race, sex, etc)."

There are other self-regulatory principles on ethics and AI, often less focused on discrimination. Examples include the Asilomar AI principles of the (US-based) Future of Life Institute, ¹⁷⁴ the Montreal Declaration for Responsible AI ¹⁷⁵ and the Principles for ethical AI of the UNI Global Union. ¹⁷⁶ IEEE, a technical professional organisation, launched a Global Initiative on Ethics of Autonomous and Intelligent Systems. ¹⁷⁷ A "Partnership on AI to Benefit People and Society" was set up by Apple, Amazon, DeepMind and Google, Facebook, IBM and Microsoft, to study and formulate best practices on AI technologies. ¹⁷⁸ In principle, such self-regulation principles are laudable. Ethical AI is obviously better than unethical AI. Self-regulatory principles could help to mitigate discrimination problems and could provide inspiration for law-makers.

-

accessed

accessed

https://www.fatml.org/resources/principles-for-accountable-algorithms 24 September 2018.

https://www.fatml.org/resources/principles-for-accountable-algorithms 24 September 2018.

https://futureoflife.org/ai-principles/ accessed 24 September 2018.

https://www.montrealdeclaration-responsibleai.com/the-declaration accessed 24 September 2018.

http://www.thefutureworldofwork.org/opinions/10-principles-for-ethical-ai/ 24 September 2018.

^{177 &}lt;u>https://standards.ieee.org/industry-connections/ec/autonomous-systems.html</u> accessed 24 September 2018. See also Koene et al. 2018.

^{178 &}lt;a href="https://www.partnershiponai.org/about/">https://www.partnershiponai.org/about/ accessed 24 September 2018. See for a list of, and critique of, other ethics principles for AI: Greene, Hoffman and Stark 2018.

However, protecting human rights cannot be left to selfregulation or soft law. 179 The main problem is that selfregulation is non-binding. Moreover, the above-mentioned principles are often somewhat abstract and do not give detailed guidance. 180 Wagner warns against "ethics washing" in the context of AI: "much of the debate about ethics seems increasingly focussed on private companies regulation. Unable or unwilling to properly provide regulatory solutions, ethics is seen as the "easy" or "soft" option which can help structure and give meaning to existing self-regulatory initiatives." 181 Indeed, self-regulation and soft law should not distract from a possible need for (hard) legal regulation. Chapter VI discusses how the law could be improved. But first we turn to recommendations to organisations using AI, and to human rights monitoring bodies and Equality Bodies.

V. RECOMMENDATIONS

What recommendations can be made on mitigating the risks of discriminatory AI, to organisations using AI, to Equality Bodies in Council of Europe member States, and to human rights monitoring bodies, such as the European Commission against Racism and Intolerance?

1. ORGANISATIONS USING AI

Several measures are important for public and private organisations wishing to prevent discrimination when they use AI. Such measures include education, obtaining technical and legal expertise, and careful planning of AI projects.

50

¹⁷⁹ See, generally on self-regulation and fundamental rights: Angelopoulos et al. 2016.

¹⁸⁰ See Campolo et al. 2017, p. 34.

¹⁸¹ Wagner 2018. See also Nemitz 2018.

Education

Education is important to make organisations realise the risks of accidental Al-driven discrimination. Relevant employees of an organisation – including managers, lawyers, and computer scientists – should be aware of the risks. As we have seen, in many examples of discriminatory AI, the organisations did not set out to discriminate. If such organisations had been aware of the risks, they might have been able to prevent that discrimination. Perhaps education could also help to mitigate the effects of "automation bias" among employees.¹⁸²

Risk assessment and mitigation

When an organisation starts an AI project, it should perform risk assessment and risk mitigation. This entails (i) involving individuals from multiple disciplines, such as computer science and law, to define the risks of a project; (ii) recording both the assessment and mitigation processes; (iii) monitoring the implementation of a project; and (iv) often reporting outward in some way, either to the public or to an oversight body. 183

Organisations should ensure that they receive help from computer scientists who understand discrimination risks. (The phrase "computer scientist" is used here as shorthand. Data scientists or other and people with sufficient knowledge of Al could also provide expertise). An emerging field in computer science focuses on discrimination risks in the field of Al decisions. Since 2014, an organisation called FATML organises workshops and conferences, with the aim of "[b]ringing together a growing community of researchers and practitioners concerned with fairness, accountability and transparency in machine learning." ¹⁸⁴ Computer scientists have published

51

.

¹⁸² Citron 2007, p. 1306. See, on automation bias: Parasuraman and Manzey 2010; Rieke, Bogen and Robinson 2018, p. 11.

¹⁸³ See eg Al Now Institute 2018; Mantelero 2018; Mantelero 2018a.

¹⁸⁴ https://www.fatml.org accessed 1 August 2018.

promising results, for instance on discrimination-aware data mining.¹⁸⁵

Defining the risks of an AI project can be challenging. When left alone, computer scientists have to make value-laden decisions while building an AI system, and often find risks or choices hard to communicate to senior decision-makers. ¹⁸⁶ Assessing and mitigating discrimination risks requires active support for those developing AI systems, and the time and money needed for this should be an active consideration in all relevant projects.

The risks and applicable legal and normative principles are different for each sector. Different risks are involved for an AI system that selects job applicants, for example, than for one that predicts crime. Therefore, experts with knowledge of a particular sector should be involved. It may be useful to set up an ethics committee to assess and discuss AI systems that entail risks for human rights. It can also be useful to bring in academics, civil society groups and potentially impacted individuals to discuss their concerns over the system.

One way to assess the risks of an AI project is to carry out an appropriate type of impact assessment. Inspiration can be drawn from the GDPR's DPIA requirement for certain risky data processing operations. 190 And organisations — especially in the public sector — should consider publishing the impact assessment report.

¹⁸⁵ See eg Custers et al. 2013; Kamiran and Calders 2009; Kamiran and Calders 2012; Kusner et al. 2017; Pedreschi, Ruggieri and Turini 2008.

¹⁸⁶ Veale, Van Kleek and Binns 2018. Kaminski 2018a, p. 30: "engineers should not be defining "discrimination" or "fairness" without extensive conversation with lawyers and community members.

¹⁸⁷ Campolo et al. 2017, p. 2.

¹⁸⁸ Council of Europe Big Data Guidelines 2017, para. 1.3.

¹⁸⁹ See Article 35(9) GDPR.

¹⁹⁰ See Council of Europe Big Data Guidelines 2017, Article 2.5. See also Reisman et al. 2018; Selbst 2017.

The risks of the AI system should also be monitored during its use, particularly as the phenomena the AI system is modelling are likely to change over time, and the risks and impacts may change with them.¹⁹¹ Organisations should consider publishing yearly reports monitoring the system.

It is often possible to prevent, or at least minimise, discriminatory effects. For instance, an organisation can choose not to use certain features as input data in their AI system. To illustrate: one US company that helps to select employees says that it does not use "distance to work" as a factor to predict which applicants will be successful employees, because that factor correlates too much with race. As reported by The Atlantic: "The distance an employee lives from work, for instance, is never factored into the score given each applicant, although it is reported to some clients. That's because different neighbourhoods and towns can have different racial profiles, which means that scoring distance from work could violate equal-employment-opportunity standards."

At Al companies and university research labs, the workforce is often not diverse – largely male and white for instance. Such organisations might pay more attention to discrimination when they have a more diverse workforce. Hence, organisations should aim to hire a more diverse workforce. ¹⁹³ Obviously, aiming for a more diverse workforce is always important.

_

¹⁹¹ See also Article 35(110 GDPR; Gama et al 2014.

¹⁹² Peck 2013. See also Rieke, Robinson and Yu 2014, p. 15.

¹⁹³ Campolo et al. 2017, p. 16.

Public sector bodies

Compared to the private sector, the public sector has extra responsibilities. Indeed, many legal rules, for instance in the field of human rights, criminal procedure law and administrative law, aim to protect people against the powerful State. The extra responsibilities also apply when public sector bodies use Al systems.

Therefore, where possible, AI systems in the public sector should be designed for transparency. ¹⁹⁴ In some situations, information about AI systems could be released to the public for scrutiny, in the spirit of the Open Data movement. Yet in some cases, such information might leak personal data and create privacy risks ¹⁹⁵ or might allow people to game the AI system. ¹⁹⁶ Therefore, public bodies might want to enable controlled access to their AI systems for researchers or civil society in secure environments, much as statistical agencies do to sensitive microdata today. ¹⁹⁷

Furthermore, public sector could adopt a sunset clause when introducing AI systems that take decisions about people. Such a sunset clause could require that a system should be evaluated, say after three years, to assess whether it brought what was hoped for. 198 If the results are disappointing, or if the disadvantages or the risks are too great, consideration should be given to abolishing the system. While public sector bodies have extra responsibilities, private sector organisations such as companies can take similar measures to those proposed above for the public sector.

¹⁹⁴ See Kroll et al. 2016; Munoz, Smith and Patil, 2016.

¹⁹⁵ Veale, Binns and Edwards 2018.

¹⁹⁶ Laskov and Lippman 2011; Bambauer and Zarsky 2018.

¹⁹⁷ See, on various degrees of openness in the open data context: Zuiderveen Borgesius, Gray and Van Eechoud 2015.

¹⁹⁸ McCray, Oye and Petersen 2010; Broeders, Schrijvers and Hirsch Ballin, p. 23.

2. EQUALITY BODIES AND HUMAN RIGHTS MONITORING BODIES

What recommendations can be made to Equality Bodies in Council of Europe member States and to human rights monitoring bodies, such as the European Commission against Racism and Intolerance, on mitigating the risks of Al-driven discrimination?

Education and technical expertise

Equality Bodies and human rights monitoring bodies should be aware of the promises and threats of AI. Therefore, education for Equality Bodies and monitoring bodies on the basics of AI and its risks is needed.

Equality Bodies and human rights monitoring bodies should also ensure that they obtain technical expertise on AI, by involving computer scientists. ¹⁹⁹ Computer scientists can recognise and understand certain risks better than, for instance, lawyers. ²⁰⁰ Computer scientists, even if they are not AI specialists, could carry out certain types of investigations into AI-driven discrimination. As Rieke, Bogen and Robinson note, "Scrutiny doesn't have to be sophisticated to be successful." ²⁰¹ Problems with an AI system can often be discovered through "simple observation of a system's inputs and outputs". ²⁰² And computer scientists who are not AI specialists themselves often know which specialists to hire for certain investigations. Depending on budget, Equality Bodies and human rights monitoring bodies could hire computer scientists for a project, or on a more permanent basis.

⁻

¹⁹⁹ As mentioned, the phrase "computer scientist" is used in this report as shorthand. Data scientists or other people with sufficient knowledge of Al could also provide expertise.

²⁰⁰ See, on the importance of technical expertise for Data Protection Authorities: Raab and Szekely 2017.

²⁰¹ Rieke, Bogen and Robinson 2018, p. 2.

²⁰² Rieke, Bogen and Robinson 2018, p. 8. They also give examples of scrutiny of Al systems (p. 31-34).

Equality Bodies and human rights monitoring bodies should consider organising public awareness campaigns for organisations in the public and private sector.²⁰³ As noted, in many cases, organisations use discriminatory AI systems by accident. Awareness could help.

More generally, schools and universities that teach computer science, data science, AI, and related topics should teach students about human rights and ethics. Many universities already offer such courses to computer science students.²⁰⁴ Equality Bodies and human rights monitoring bodies could consider assisting schools and universities with such courses.²⁰⁵

To permit public debate, it would be good if the general public knew more about the risks of discriminatory AI – and about the many advantages and possibilities of AI. However, awareness building should not lead to responsibilisation. This term describes "the process whereby subjects are rendered individually responsible for a task which previously would have been the duty of another – usually a state agency – or would not have been recognized as a responsibility at all." ²⁰⁶ Policymakers should not make people responsible for defending themselves against discrimination. ²⁰⁷ That said, awareness is important for an inclusive debate on the risks of AI decisions.

⁻

²⁰³ See ECRI Statute Resolution 2002, Article 12; ECRI general policy recommendation no. 2 (2018), para. 13(e); para. 34, and explanatory memorandum para. 64.

²⁰⁴ Fiesler 2018 compiled a list of more than 200 courses on tech ethics.

²⁰⁵ See ECRI general policy recommendation no. 10: on combating racism and racial discrimination in and through school education, 15 December 2006, Strasbourg, CRI(2007)6 https://rm.coe.int/ecri-general-policy-recommendation-no-10-on-combating-racism-and-racia/16808b5ad5 accessed 14 October 2018.

²⁰⁶ Wakefield and Fleming 2009.

²⁰⁷ See also Ellis and Watson 2012, p. 502-503.

Prior consultation with Equality Bodies

Equality Bodies could require public sector bodies to discuss with them any planned projects that involve AI decision-making about individuals or groups. For instance, an Equality Body could help to assess whether training data are biased. ²⁰⁸ Equality Bodies could also require each public sector body using AI decision-making about people to ensure that it has sufficient legal and technical expertise to assess and monitor risks. And public sector bodies could be required to regularly assess whether their AI systems have discriminatory effects. (Depending on the national situation, Equality Bodies could also suggest, rather than require).

Equality Bodies and human rights monitoring bodies could help to develop a specific method for a "human rights and AI impact assessment". As mentioned, impact assessments can be useful – but to date, there is no specific impact assessment method for AI.²⁰⁹ When developing such a method, different stakeholders and people from different disciplines should be involved. Inspiration can be drawn from privacy and data protection impact assessments.²¹⁰

Engage in public procurement processes

Equality Bodies should seek, through national provisions and processes as well as through lobbying for increased access, to be involved in the procurement of public-sector AI systems from an early stage. Equality Bodies can help ensure that concerns around discrimination are built into the AI systems being

_

²⁰⁸ See ECRI general policy recommendation no. 2 (2018), Article 13(g). See also Article 36 GDPR, on prior consultation.

²⁰⁹ Reisman et al. 2018 discuss "algorithmic impact assessments" in the US. But since Council of Europe member States have different legal systems than the US, a US method can provide inspiration, but cannot be directly applied here.

²¹⁰ See Binns 2017; Kloza et al. 2017; Mantelero 2017; Wright and De Hert 2012. See also the Brussels Laboratory for Data Protection and Privacy Impact Assessments http://dpialab.org/.

procured: that systems are open enough to audit and subject to appropriate safeguards.

Cooperate with Data Protection Authorities

As said, for Al-driven discrimination, the two most relevant legal frameworks are non-discrimination law and data protection law. It would be a shame if those fields of law operate in their own silos.²¹¹ Equality Bodies should cooperate with Data Protection Authorities. For instance, it could be helpful to exchange knowledge and to learn from one another's experiences. 212 Many Data Protection Authorities have some technical expertise in house, 213 and some have experience with hiring outside computer scientists for research projects. 214 Data Protection Authorities may learn about organisations that use Al systems that entail discrimination risks, and could warn Equality Bodies. Equality Bodies could provide information to Data Protection Authorities, for instance about discrimination risks. Depending on the national situation, it could also be useful for Equality Bodies to cooperate with Consumer Protection Authorities and Competition law authorities.

For cooperation and knowledge sharing between different types of regulators, the European Data Protection Supervisor proposed in 2016 to set up "a voluntary network of regulatory bodies to share information (...) about possible abuses in the digital ecosystem and the most effective way of tackling

²¹¹ See Schreurs et al. 2008; Gellert et al. 2013; Hacker 2018; Lammerant, De Hert, Blok 2017.

²¹² See ECRI general policy recommendation no. 2 (2018), Article 13(b).

²¹³ Raab and Szekely 2017.

²¹⁴ For instance, researchers of the University of Leuven have examined Facebook's tracking for the Data Protection Authority in Belgium. See Belgian Data Protection Authority 2018.

them."²¹⁵ Perhaps that initiative could provide inspiration for Equality Bodies and human rights monitoring bodies.²¹⁶

Cooperate with academics

Equality Bodies and human rights monitoring bodies should keep in touch with, and perhaps cooperate with, academics. This report illustrates how many examples of discriminatory AI decisions were discovered by academic researchers (and by investigative journalists). 217 Many academics love to assist regulators but are not in regular contact with them. In the short term, Equality Bodies and monitoring bodies could visit conferences and other events where academic researchers meet. At many international privacy conferences, discriminatory Al is a much-debated topic. Several of these conferences attract a mix of regulators, practitioners, civil society groups and scholars from different disciplines, such as law, computer science, philosophy and sociology, 218 Equality Bodies and monitoring bodies could also consider organising conferences, round tables or other events on discrimination risks of AI, to foster contacts between the research community and Equality Bodies. And perhaps Equality Bodies and monitoring bodies could commission more research on Al's discrimination risks

_

²¹⁵ European Data Protection Supervisor 2016.

²¹⁶ As an aside: within universities too, more cooperation is needed between different types of legal scholars, such as non-discrimination law specialists (often working at human rights institutes) and data protection law specialists (often working at law and technology institutes).

²¹⁷ See Rieke, Bogen and Robinson 2018, p. 31.

²¹⁸ See for instance: the CPDP Computers, Privacy and Data Protection conference in Brussels https://www.cpdpconferences.org; the APC Amsterdam Privacy Conference https://www.apc2018.com; TILTing Perspectives https://www.tilburguniversity.edu/research/institutes-and-research-groups/tilt/events/tilting-perspectives; and the PLSC Privacy Law Scholars Conference https://www.fatml.org. Accountability, and Transparency (ACM FAT*) will be in Amsterdam in 2020: https://www.fatml.org. All accessed 14 October 2018.

(see section VI.3) or set up a working party on Al's discrimination risks.²¹⁹

Equality Bodies and human rights monitoring bodies should not only engage with civil society groups that work on discrimination ²²⁰ but also with consumer groups ²²¹ and civil society groups that focus on technology policy and digital rights. ²²² Civil society groups that work on discrimination often have different expertise from groups that work on technology and digital rights. More contact between such groups would be useful too, as many of them are interested in Al-driven discrimination. ²²³

Litigation and regulation

Depending on the national situation, Equality Bodies could also engage in strategic litigation in the area of AI decision-making.²²⁴ And Equality Bodies and human rights monitoring bodies could push for regulation to mitigate discrimination risks of AI.²²⁵ Suggestions to improve regulation are discussed in the next chapter.

²¹⁹ See ECRI Statute Resolution 2002, Article 6(1); 6(2); ECRI general policy recommendation no. 2 (2018), article 13(d).

²²⁰ See ECRI Statute Resolution 2002, Article 10(1) and 13.

²²¹ For consumer organisations, BEUC (the European Consumer Organisation) could be a point of contact. BEUC's members are 43 consumer organisations from 32 European countries. https://www.beuc.eu/about-beuc/who-we-are accessed 10 October 2018. See also European Consumer Organisation BEUC 2018.

²²² For groups focusing on rights and freedoms in the digital environment, European Digital Rights (EDRi) could be a point of contact. EDRi is an association of civil and human rights organisations from across Europe. https://edri.org/members/ accessed 10 October 2018.

²²³ See Gangadharan and Niklas 2018, who interviewed NGOs and conclude that better cooperation is needed between (i) privacy- and technology-oriented NGOs and (ii) discrimination-oriented NGOs.

²²⁴ See ECRI general policy recommendation no. 2 (2018), Article 14-16.

²²⁵ See ECRI Statute Resolution 2002, Article 1; ECRI general policy recommendation no. 2 (2018), Article 13(j).

VI. IMPROVING REGULATION

Which types of action (legal, regulatory, self-regulatory) can reduce risks?

Current law has weaknesses when applied to Al-driven discrimination, as we saw in chapter IV. Additional regulation is probably needed to protect people against illegal discrimination and unfair differentiation. Section 1 provides preliminary remarks about regulating in the area of fast-developing technology. Section 2 focuses on improving enforcement of existing non-discrimination norms. Section 3 discusses whether the legal norms themselves should be amended because of Al decision-making. The suggestions in this chapter are meant as starting points for discussion rather than as definitive policy advice.

1. REGULATION AND FAST-DEVELOPING TECHNOLOGY

Regulating brings extra challenges when the rules are to apply to fast-developing technology. Adopting statutes or treaties may take years or even decades. Meanwhile, technology, the market and society develop quickly.

These challenges are not unique for AI; there is experience with regulating new technologies. When regulating in the area of new technologies, policy-makers can combine different types of rules, such as statutes with broad principles and guidelines (by regulators for instance) with more specific rules. The statutes could be phrased in a reasonably technology-neutral way. Technology-neutral legal provisions with broad principles have the advantage of not having to be changed every time a new technology is developed. A disadvantage is that broad principles can be difficult to apply in practice. Therefore,

-

²²⁶ See Koops 2006.

guidance by regulators can be useful. 227 Guidelines can be amended faster and can thus be more specific and concrete. Guidelines should be evaluated regularly and amended whenever required. 228

Data protection law partly takes this combined approach. ²²⁹ Data protection law (such as the GDPR and the modernised Convention 108) contains many broadly phrased provisions that can be applied to different situations and technologies. ²³⁰ For instance, data protection law does not contain specific rules for CCTV, or for monitoring in the workplace. But as far as personal data (including on video images) are used, data protection law does apply to CCTV and workplace monitoring.

In addition to data protection law's statutory provisions, Data Protection Authorities often adopt interpretative guidelines with more specific and concrete requirements for different situations, such as CCTV, ²³¹ the workplace ²³² and automated decision-making. ²³³ In the EU, the European Data Protection Board and its predecessor have adopted more than 250 guidelines since 1995. ²³⁴ Similarly, the Council of Europe has adopted

_

²²⁷ See Zuiderveen Borgesius 2015a, chapter 9, section1; Baldwin, Cave and Lodge 2011, chapter 14.

²²⁸ See Koops 2006.

²²⁹ I am not suggesting that data protection law should be seen as a best practice for regulating in fields where technology develops quickly. There is plenty to criticise in data protection law.

²³⁰ Data protection law, developed since the early 1970s, could itself be seen as the legal answer to a new development: large-scale bureaucracies and automated personal data processing. Since its inception, data protection law has been adapted continuously to new developments – as illustrated by the recent GDPR and Modernised Convention 108.

²³¹ Article 29 Working Party 2004 (WP89).

²³² Article 29 Working Party 2017 (WP249).

²³³ Article 29 Working Party 2018 (WP251).

²³⁴ See the website of the European Data Protection Board https://edpb.europa.eu/edpb_en. Its predecessor was called the Article 29 Working Party https://ec.europa.eu/newsroom/article29/news.cfm?item_type=1308. The opinions and guidelines are also compiled on this site: <a href="https://iapp.org/resources/article/all-of-the-european-particle/all-of-the-

guidelines in addition to the Data Protection Convention 108, for instance on big data, ²³⁵ the police sector ²³⁶ and profiling. ²³⁷

Hence, if new legal rules were adopted to mitigate discrimination risks in the area of AI, perhaps statutory rules should be combined with a possibility for regulatory bodies to adopt guidelines that are easier to amend. There are more possibilities than statutory law and regulator guidance, such as co-regulation: self-regulation with varying degrees of influence of public regulators. The basic idea remains the same: different types of rules can be combined.²³⁸ As Koops puts it, "Through multi-level legislation, open-ended formulations and a mixed approach of abstract and concrete rules that are periodically evaluated, adequate legal certainty with respect to current technologies may be ensured, while at the same time sufficient scope is given for future technological developments."²³⁹

Of course, there must be democratic legitimacy and sufficient checks and balances regarding entities that set rules or guidelines. In sum, regulating in the area of new technologies is hard, but possible – and often necessary.

<u>data-protection-board-and-article-29-working-party-guidelines-opinions-and-documents/,</u> links accessed on 10 October 2018.

²³⁵ Council of Europe Big Data Guidelines 2017.

²³⁶ Council of Europe Police and Personal Data Guide 2018.

²³⁷ Council of Europe, Profiling Recommendation 2010.

²³⁸ See Angelopoulos et al. 2016, p. 5-6; Brown and Marsden 2013. Specifically on coregulation: Hirsch 2010; Kaminsky 2018a.

²³⁹ Koops 2006.

2. ENFORCEMENT

Improving enforcement of current non-discrimination norms

Regarding discrimination in the area of AI-driven decisions, the overarching norms are reasonably clear – in our society we do not, and should not, accept discrimination on the basis of protected characteristics such as racial origin. Below are some suggestions on enforcement of non-discrimination norms in the area of AI.

Transparency

As noted, one of the problems with AI systems is the lack of transparency; their "black box" character.²⁴⁰ The opaqueness can be seen as a problem in itself – but the opaqueness also makes it harder to discover discrimination.

Regulation can aim to improve transparency. The law (including guidelines etc) could, for instance, require that AI systems used in the public sector are developed in such a way that they enable auditing and explainability.²⁴¹ For the private sector too, such requirements could be considered. ²⁴² There are precedents for such requirements in the private sector; a requirement of interpretability exists for certain systems for algorithmic trading.²⁴³

For some types of systems, it could be useful if public sector bodies release the underlying code (software). Sometimes, examining the code can provide information about how a system works. As Rieke, Bogen, and Robinson note, "code audits are most likely to be useful when there is a clearly defined question about how a software program operates in regulated

²⁴⁰ Pasquale 2015. See also Zarsky 2018.

²⁴¹ See Rieke, Bogen and Robinson 2018, p. 6; Pasquale 2017. See, on auditing Al systems: Sandvig et al. 2014.

²⁴² See Rieke, Bogen and Robinson 2018, p. 6.

²⁴³ See section IV.3; the part about algorithmic trading.

space, and particular standards against which to measure a system's behaviour or performance."²⁴⁴ Freedom of information laws could be adapted so that the code in AI systems is subject to such laws. Such an amendment would enable journalists, academics and others to obtain and examine such code.

Al systems are often protected by trade secrets, intellectual property rights or a company's terms and conditions. Such protection makes it harder for regulators, journalists, and academics to investigate such systems. Perhaps the law should be adapted to improve research exceptions and to enable some types of research. And perhaps the law should require organisations to disclose certain information to researchers upon request. Such regulation must strike a delicate balance between public interest in transparency and commercial, privacy and other interests in opaqueness.

In many cases, the code alone does not give much information about an AI system, as the system can only be assessed when it is used in practice. "For even moderately complex programs," observe Rieke, Bogen, and Robinson, "it may be necessary to see a program run "in the wild," with real users and data to truly understand its effects."²⁴⁷

The law could require the public sector to use only AI systems that have been properly assessed for risks and enable oversight and auditing.²⁴⁸ A similar requirement could be considered for the private sector when AI systems are used for certain decisions, for instance on eligibility for insurance, credit or a job.²⁴⁹ More research and debate is needed on who should

²⁴⁴ Rieke, Bogen and Robinson 2018, p. 19.

²⁴⁵ See Bodo et al. 2017, p. 171-175; Malgieri 2016; Wachter and Mittelstadt 2018, p. 63-77.

²⁴⁶ Similar questions arise in open data versus privacy discussions. See Zuiderveen Borgesius, Gray and Van Eechoud 2015.

²⁴⁷ Rieke, Bogen and Robinson 2018, p. 19.

²⁴⁸ See Campolo et al. 2018, p. 1.

²⁴⁹ See Campolo et al. 2018, p. 1.

conduct such audits. For oversight and auditing of AI systems, an organisation needs considerable expertise.²⁵⁰

Investigation and enforcement powers

Council of Europe member States should ensure that Equality Bodies and Data Protection Authorities receive adequate funding, and that they have sufficient investigation and enforcement powers.²⁵¹ Without enforcement, transparency will not necessarily lead to accountability.²⁵²

In sum, Equality Bodies and human rights monitoring bodies can push for regulation that enables better enforcement of current non-discrimination norms in the area of Al decision-making. However, Al decision-making also opens the way for new types of discrimination and differentiation that largely escape current non-discrimination and other laws. We turn to that topic now.

3. REGULATING NEW TYPES OF DIFFERENTIATION

Non-discrimination law and data protection law leave gaps in the context of AI.²⁵³ Many non-discrimination statutes apply only to certain protected characteristics, such as race, gender or sexual orientation.²⁵⁴ The statutes do not apply to discrimination on the basis of financial status for instance. Data protection law can help to fill some, but definitely not all, gaps in non-discrimination law.

Al systems can escape non-discrimination law when they differentiate on the basis of newly invented classes.²⁵⁵ To give a simplified example: suppose an Al system finds a correlation

²⁵⁰ It has been suggested that a specific oversight body for automated profiling (Al-driven decision-making) might be useful. See Koops 2008.

²⁵¹ See ECRI general policy recommendation no. 2 (2018), Article 28.

²⁵² See Kaminski 2018a, p. 21.

²⁵³ See section IV.1 and IV.2.

²⁵⁴ Gerards 2007: Khaitan 2015.

²⁵⁵ Custers 2004. See also Mittelstadt et al. 2016.

between (i) using a certain web browser and (ii) a greater willingness to pay. An online shop could charge higher prices to people using that browser. ²⁵⁶ Such practices would remain outside the scope of non-discrimination law, as a browser type is not a protected characteristic. (For this hypothesis we assume that the browser type is not a proxy for a protected characteristic).

Al can reinforce social inequality

But AI decisions that remain outside the scope of nondiscrimination law can still lead to differentiation that is unfair or has other drawbacks. For instance, insurance companies could use AI systems to set premiums for individual consumers, or to deny some consumers insurance. To some extent, risk differentiation is necessary, and an accepted practice, for insurance. And it can be considered fair when high-risk customers pay higher premiums.

But there are drawbacks. Too much risk differentiation could make insurance unaffordable for some consumers and could threaten the risk-pooling function of insurance. Furthermore, risk differentiation might result in the poor paying more. A consumer who lives in a poor neighbourhood with many burglaries might pay more for house insurance, because the risk of a burglary is higher. But if neighbourhoods where many poor people live have higher risks, then poor people pay, on average, more ²⁵⁷

²⁵⁶ There is no evidence of such practices, although from a technical perspective such price discrimination is easy. There was, however, a travel and booking site that showed more expensive hotels to Apple users and cheaper ones to PC users (Mattioli 2012).

²⁵⁷ See on Al and insurance: Dutch Association of Insurers 2016; Financial Conduct Authority 2016; Peppet 2014; Swedloff 2014. Germany has a specific rule on automated decisions in the insurance context. See Bundesdatenschutzgesetz vom 30. Juni 2017 (BGBI. I S. 2097), Section 37 https://www.gesetze-im-internet.de/englisch_bdsg/englisch_bdsg.html#p0310 accessed 13 October 2018. See also Malgieri 2018, p. 9-11. On discrimination and insurance: Avraham 2017

More generally, Al could reinforce social inequality. For instance, Valentino-De Vries, Singer-Vine and Soltani showed that some online price differentiation practices in the US had the effect that people in poor areas paid higher prices. Several shops charged more to consumers who live in the countryside than to consumers in large cities. ²⁵⁸ In the countryside, consumers have to drive hours to visit a competitor. Therefore, an online shop does not have to use cheap prices; most customers will not drive for hours to buy the product at a cheaper price. In a large city, a consumer can easily go to a competitor to buy a product. Therefore, some online shops offered cheaper prices in large cities. This pricing scheme had the effect, probably unintentionally, that poorer people paid, on average, higher prices, as people tend to be poorer in the countryside of the US.²⁵⁹ AI can thus reinforce social inequality. But, as noted, someone's financial status is not a protected characteristic, so non-discrimination law does not regulate such a practice (assuming that the practice does not lead to indirect discrimination based on a protected characteristic).²⁶⁰

Al can lead to errors

Non-discrimination law has little to say about incorrect Al predictions (false positives and false negatives). A problem with Al decisions is that they are often incorrect for a particular individual. Al decision-making often entails applying a predictive model to individuals. A simplified example of a predictive model is: "80% of the people living in postal code F-67075 pay their bills late." If, based on this group profile, a company denies loans to all people in postal code F-67075, it

_

²⁵⁸ Valentino-De Vries, Singer-Vine and Soltani 2012.

²⁵⁹ Valentino-De Vries, Singer-Vine and Soltani 2012. See on reinforcing inequality and "social sorting" also Atrey 2018; Danna and Gandy 2002; Lyon 2002; Naudts 2017; Taylor 2017; Turow 2011. Gandy warned 25 years ago for the discriminatory effects of large-scale data processing (Gandy 1993).

²⁶⁰ And in data protection law, data about somebody's financial status is not among the "special categories" of data (Article 9 GDPR).

also denies loans to the 20% who pay their bills on time.²⁶¹ Such practices could disproportionately harm certain groups in society. Sometimes, an Al system makes more errors for minority groups than for the majority.²⁶²

New rules?

Additional regulation should be considered, because Al decision-making that escapes non-discrimination law can still be unfair. But it is probably not useful to adopt rules for Al decision-making in general. All is used in many different sectors and for many purposes, and often, All does not threaten human rights.²⁶³ An All system of a chess computer does not bring the same risks as an All system for predictive policing.

Even for AI systems that make decisions about humans, the risks are different in different sectors, and different rules should apply. The fairness of AI decision-making cannot be assessed in the abstract. In each sector, or application area, different arguments have different weights.²⁶⁴ And in different sectors, different normative and legal principles apply. For instance, the right to a fair trial and the presumption of innocence are important in the field of criminal law. In consumer transactions, freedom of contract is an important principle. Hence, when new rules are considered, such rules need to focus on specific sectors.

Whether there is a need for new rules could be assessed as follows. For a particular sector, several questions should be answered.

(i) Which rules apply in this sector, and what are the rationales for those rules? A rule may, for example, aim

²⁶² See, for an example of a system with more errors for minorities: Rieke, Robinson and Yu 2014, p. 12. In such a situation, the Al-driven decisions could be a form of prohibited indirect discrimination. See also Hardt 2014.

²⁶¹ Zarsky 2002.

²⁶³ See Royal Society 2017, p. 99.

²⁶⁴ Schauer 2003. See also Wachter and Mittelstadt 2018, p. 83.

to protect a human right, or express a legal principle, such as equality, contractual freedom, or the right to a fair trial. Economic rationales also differ from sector to sector. For instance, risk pooling is important for insurance, while it is not relevant in most other sectors. Hence, for each sector the rationales behind the rules differ.

- (ii) How is or could AI decision-making be used in this sector, and what are the risks? For instance, false positives are a serious problem in the context of criminal law. A false positive could lead to people being questioned, arrested or perhaps even punished. We should not accept AI decision-making that breaches the underlying values of criminal law. By contrast: if an incorrect decision by an AI system for price discrimination makes a consumer pay extra, the effect is often less harmful than when an incorrect AI decision leads to someone being arrested by the police.
- (iii) Considering the rationales for the rules in this sector, should the law be improved in the light of AI decisionmaking? Does AI threaten the law's underlying principles or undermine the law's goals? If current law leaves important risks unaddressed, amendments should be considered.

In conclusion, new rules may be needed for Al decision-making, to protect fairness and human rights such as the right to non-discrimination. However, more research and debate are required on the questions of whether and which rules are needed.

Empirical and technical research

Information is necessary for good policy. There is a clear need for more information about Al-driven discrimination, and hence for more research. Council of Europe member States should support research – research by human rights monitoring bodies, Equality Bodies, and by academics. More empirical research is needed for instance. It is unclear on what scale Al decision-making is used. How often does algorithmic decision-making lead to discrimination (on the basis of racial origin for instance)? And to other types of unfair differentiation?

More computer science research into solutions is needed too. For instance, how could AI systems be designed so they respect and promote human rights, fairness and accountability? Can training data be checked for discrimination risks?²⁶⁶ As noted, an emerging and vibrant field of computer science focuses on such questions.²⁶⁷ More generally: if countries fund AI research, part of that funding should be used for research into the risks for fairness and human rights, and into mitigating those risks.

Normative and legal research

There is also a need for public debate, and for normative and legal research. How could the prohibition of indirect discrimination be enforced more effectively? How should the law deal with unfair differentiation that remains outside the scope of non-discrimination law? How to define fairness in diverse sectors? How should the law (and technology) protect

²⁶⁵ See Wagner et al. 2018, p. 43.

²⁶⁶ See Campolo et al. 2018, p. 1

²⁶⁷ https://www.fatml.org accessed 2 October 2018.

people against intersectional²⁶⁸ and structural discrimination?²⁶⁹ Should the law protect some types of "group privacy", and how?²⁷⁰ How to safeguard the rule of law when AI systems make decisions about people?²⁷¹ Which types of decisions, if any, should never be taken by computers? How could data protection law be used in practice to fight discrimination? Are new rules needed, or are tweaks to non-discrimination law and data protection law sufficient? Which tweaks would be needed? Which new rules would be needed?

VII. CONCLUSION

In conclusion, AI offers many exciting possibilities to improve our societies. But AI decision-making also brings risks – it is often opaque and can have discriminatory effects, for instance when an AI system learns from data reflecting biased human decisions.

In the public and the private sector, organisations can take Aldriven decisions with far-reaching effects for people. Public sector bodies can use AI for predictive policing or sentencing recommendations, and for decisions on, for instance, pensions, housing assistance or unemployment benefits. The private sector can also take AI decisions with major consequences for people, such as decisions regarding employment, housing or credit. Moreover, many small decisions, taken together, can have large effects. One targeted advertisement is rarely a major problem, but when aggregated, targeted advertising may exclude some groups. And AI-driven price differentiation could lead to certain groups in society consistently paying more.

⁻

²⁶⁸ See on intersectional discrimination: Crenshaw 1989; Fredman 2016. See also ECRI General policy recommendation no. 14 on combating racism and racial discrimination in employment, adopted on 22 June 2012, CRI(2012)48, https://rm.coe.int/ecri-general-policy-recommendation-no-14-on-combating-racism-and-racia/16808b5afc accessed 14 October 2018.

²⁶⁹ See on structural discrimination: para. 20 of the explanatory memorandum of ECRI general policy recommendation no. 2 (2018).

²⁷⁰ See Bygrave 2002, chapters 9-16; Taylor, Van der Sloot, and Floridi 2017; Vedder 1997.

²⁷¹ Hildebrandt 2015; Bayamlıoğlu and Leenes 2018.

The most relevant legal instruments to mitigate the risks of Aldriven discrimination are non-discrimination law and data protection law. If effectively enforced, both legal instruments could help to fight illegal discrimination. Council of Europe member States, human rights monitoring bodies, such as the European Commission against Racism and Intolerance, and Equality Bodies should aim for better enforcement of current non-discrimination norms.

But Al also paves the way for new types of unfair differentiation (or discrimination) that escape current laws. Most nondiscrimination statutes only apply to discrimination on the basis of protected characteristics, such as racial origin. Such statutes do not apply if organisations differentiate on the basis of newly invented classes that do not correlate with protected characteristics. Such differentiation could still be unfair. however, for instance when it reinforces social inequality. We probably need additional regulation to protect fairness and human rights in the area of AI. But regulating AI in general is not the right approach, as the use of Al systems is too varied for one set of rules. We need sector-specific rules, because different values are at stake, and different problems arise, in different sectors. More debate and interdisciplinary research are needed. If we make the right choices now, we can enjoy the many benefits of AI, while minimising the risks of unfair discrimination.

BIBLIOGRAPHY

A shared statement of civil rights concerns 2018, 'The use of pretrial risk assessment instruments. A shared statement of civil rights concerns', signed by 119 civil rights organisations in the US' (2018)

http://civilrightsdocs.info/pdf/criminal-justice/Pretrial-Risk-Assessment-Full.pdf accessed on 4 October 2018.

Allen A, 'The 'three black teenagers' search shows it is society, not Google, that is racist'. 10 June 2016. The Guardian.

https://www.theguardian.com/commentisfree/2016/jun/10/three-black-teenagers-google-racist-tweet accessed 1 October 2018.

Alpaydin E, Machine learning: the new AI (MIT Press 2016).

Ananny M and Crawford K, 'Seeing without knowing: Limitations of the transparency ideal and its application to algorithmic accountability' (2018) 20(3) New Media & Society 973.

Angelopoulos, C. et al. 'Study of fundamental rights limitations for online enforcement through self-regulation, report IVIR Institute for Information Law, University of Amsterdam' (2016)

https://www.ivir.nl/publicaties/download/1796 accessed 1 October 2018.

Angwin J, Scheiber N and Tobin A, 'Dozens of companies are using Facebook to exclude older workers from job ads', ProPublica, December 2017 https://www.propublica.org/article/facebook-ads-age-discrimination-targeting accessed 19 September 2018.

Angwin J, Tobin A and Varner M, 'Facebook (still) letting housing advertisers exclude users by race', ProPublica, November 2017 https://www.propublica.org/article/facebook-advertising-discrimination-housing-race-sex-national-origin accessed 19 September 2018.

Angwin J et al., 'Machine bias: There's software used across the country to predict future criminals. And it's biased against blacks'. ProPublica, May 23, 2016 https://www.ProPublica.org/article/machine-bias-risk-assessments-in-criminal-sentencing accessed 19 September 2018.

Angwin J, Mattu S and Larson J, 'The tiger mom tax: Asians are nearly twice as likely to get a higher price from Princeton Review' (2015) ProPublica. https://www.ProPublica.org/article/asians-nearly-twice-as-likely-to-get-higher-price-from-princeton-review.

Article 29 Working Party, 'Opinion 4/2004 on the Processing of Personal Data by means of Video Surveillance' (WP 89), 11 February 2004.

Article 29 Working Party, 'Guidelines on Data Protection Impact Assessment (DPIA) and determining whether processing is "likely to result in a high risk"

for the purposes of Regulation 2016/679' (WP 248 rev.01), Brussels, 4 October 2017.

Article 29 Working Party, 'Opinion 2/2017 on data processing at work' (WP249), 8 June 2017.

Article 29 Working Party, 'Guidelines on Automated individual decision-making and Profiling for the purposes of Regulation 2016/679' (WP251rev.01), 6 February 2018.

Atrey S, 'The intersectional case of poverty in discrimination law' (2018) 18(3) Human Rights Law Review 411.

Avraham R, 'Discrimination and Insurance' in Lippert-Rasmussen, Kasper (ed), The Routledge handbook of the ethics of discrimination (Routledge 2017).

Bagli, CV, 'Facebook vowed to end discriminatory housing ads. Suit says it didn't', New York Times 27 March 2018,

https://www.nytimes.com/2018/03/27/nyregion/facebook-housing-ads-discrimination-lawsuit.html accessed 14 October 2018.

Baldwin R, Cave M and Lodge M, Understanding regulation: Theory, strategy, and practice (2nd edition) (Oxford University Press 2011).

Bambauer JR and Zarsky T, 'The algorithm game', 7 March 2018, Notre Dame Law Review (2018 Forthcoming) https://ssrn.com/abstract=3135949 accessed 12 October 2018.

Barocas S and Selbst AD, 'Big Data's disparate impact' (2016) 104 Calif Law Rev 671.

Bayamlıoğlu E and Leenes R, 'The "rule of law" implications of data-driven decision-making: a techno-regulatory perspective' (2018) DOI: 10.1080/17579961.2018.1527475 Law, Innovation and Technology.

BBC News, 'Google apologises for Photos app's racist blunder', 1 July 2015, https://www.bbc.com/news/technology-33347866 accessed 1 October 2018.

Belgian Data Protection Authority, 'Victory for the authority in Facebook proceeding', 16 February 2018,

https://www.dataprotectionauthority.be/news/victory-privacy-commission-facebook-proceeding accessed 2 October 2018.

Binns R, 'Data protection impact assessments: A meta-regulatory approach' (2017) 7(1) International Data Privacy Law 22.

Binns et al., "It's reducing a human being to a percentage"; Perceptions of justice in algorithmic decisions", CHI 2018, April 21–26, 2018, Montréal, QC, Canada, https://dl.acm.org/citation.cfm?id=3173951 accessed 1 October 2018.

Bodo B et al., 'Tackling the algorithmic control crisis-the technical, legal, and ethical challenges of research into algorithmic agents' (2017) 19 Yale JL & Tech. 133.

Broeders D, Schrijvers E and Hirsch Ballin E, 'Big data and security policies: serving security, protecting freedom, WRR-Policy Brief 6, Netherlands Scientific Council for Government Policy (WRR)' (2017)

https://www.wrr.nl/binaries/wrr/documenten/policy-briefs/2017/01/31/bigdata-and-security-policies-serving-security-protecting-

<u>freedom/WRR PB6 BigDataAndSecurityPolicies.pdf</u> accessed 1 October 2018, p. 24-25.

Bryson J, 'Three very different sources of bias in AI, and how to fix them', Adventures in NI, 13 July 2017, <a href="https://joanna-nt/miss.com/https://ioanna-nt/miss.com/https://ioa

bryson.blogspot.com/2017/07/three-very-different-sources-of-bias-in.html, accessed 11 October 2018.

boyd dm and Crawford K, 'Critical questions for big data: Provocations for a cultural, technological, and scholarly phenomenon' (2012) 15(5) Information, Communication & Society 662.

Brown I and Marsden CT, Regulating code: Good governance and better regulation in the information age (MIT Press 2013).

Buolamwini J and Gebru T, 'Gender shades: Intersectional accuracy disparities in commercial gender classification' (Conference on Fairness, Accountability and Transparency 2018) 77.

Burrell J, 'How the machine 'thinks': understanding opacity in machine learning algorithms', Big Data & Society (2016) 3(1). p. 1-12.

Bygrave LA, 'Minding the machine: Article 15 of the EC Data Protection Directive and automated profiling' (2001) 17 Computer Law & Security Report 17.

Bygrave LA, Data protection law: approaching its rationale, logic and limits (Information Law Series, Kluwer Law International 2002).

Cabañas JG, Cuevas Á and Cuevas R, 'Facebook use of sensitive data for advertising in Europe' (2018) arXiv preprint arXiv:1802.05030.

Caliskan A, Bryson JJ and Narayanan A, 'Semantics derived automatically from language corpora contain human-like biases' (2017) 356(6334) Science 183.

Campolo A et al, 'Al Now 2017 Report' (2017) https://ainowinstitute.org/Al_Now_2017_Report.pdf accessed 1 October 2018.

Chouldechova A, 'Fair prediction with disparate impact: A study of bias in recidivism prediction instruments' (2017) 5 Big Data 153 http://dx.doi.org/10.1089/big.2016.0047 accessed 10 October 2018.

Citron DK, 'Technological due process' (2007) 85 Wash.UL Rev. 1249.

Cobbe J, 'Administrative Law and the machines of government: Judicial review of automated public-sector decision-making' (2018) Presented at the Microsoft Cloud Computing Research Centre 2018 Symposium, University of Cambridge. Available on SSRN

https://papers.ssrn.com/sol3/papers.cfm?abstract_id=3226913 accessed 4 October 2018.

Collins H and Khaitan T, 'Indirect discrimination law: Controversies and critical questions' in Collins, H. and T. Khaitan (eds), Foundations of Indirect Discrimination Law (Hart Publishing 2018).

Council of Europe Big Data Guidelines 2017: Consultative Committee of the Convention for the Protection of Individuals with regard to Automatic Processing of Personal Data. 'Guidelines on the protection of individuals with regard to the processing of personal data in a world of Big Data', Strasbourg, 23 January 2017, T-PD(2017)01, https://rm.coe.int/16806ebe7a accessed 1 October 2018.

Council of Europe MSI-AUT 2018, Council of Europe Committee of experts on Human Rights Dimensions of automated data processing and different forms of artificial intelligence, https://www.coe.int/en/web/freedom-expression/msi-aut#{"32639232":[0]} accessed 30 September 2018.

Council of Europe Police and Personal Data Guide 2018, Consultative Committee of the Convention for the Protection of Individuals with regard to Automatic Processing of Personal Data, 'Practical guide on the use of personal data in the police sector', T-PD(2018)01, 15 February 2018.

Council of Europe, Profiling recommendation 2010, 'Recommendation CM/Rec(2010)13 of the Committee of Ministers to member states on the protection of individuals with regard to automatic processing of personal data in the context of profiling' (Adopted by the Committee of Ministers on 23 November 2010 at the 1099th meeting of the Ministers' Deputies).

Crawford K, 'Think again: Big data', Foreign Policy 10 May 2013 https://foreignpolicy.com/2013/05/10/think-again-big-data/ accessed on 10 October 2018.

Crenshaw K, 'Demarginalizing the intersection of race and sex: A black feminist critique of antidiscrimination doctrine, feminist theory and antiracist politics' (1989) U.Chi.Legal F. 139.

Custers B, The power of knowledge, Ethical, legal, and technological aspects of data mining and group profiling in epidemiology (Wolf Legal Publishers 2004).

Custers B; Calders T; Schermer BW; Zarsky TZ, Discrimination and Privacy in the Information Society (Springer 2013).

Dalenberg DJ, 'Preventing discrimination in the automated targeting of job advertisements' (2018) 34(3) Computer Law & Security Review 615.

Danna A and Gandy Jr OH, 'All that glitters is not gold: Digging beneath the surface of data mining' (2002) 40(4) J Bus Ethics 373.

Dastin J, 'Amazon scraps secret AI recruiting tool that showed bias against women', 10 October 2018. Reuters, https://www.reuters.com/article/us-amazon-com-jobs-automation-in...-ai-recruiting-tool-that-showed-bias-against-women-idUSKCN1MK08G accessed 10 October 2018.

Datta A, Tschantz MC and Datta A, 'Automated experiments on ad privacy settings' (2015) 2015(1) Proceedings on Privacy Enhancing Technologies 92.

Datta A et al., 'Discrimination in online advertising: A multidisciplinary inquiry' (Conference on Fairness, Accountability and Transparency 2018) 20.

De Hert P and Gutwirth S, 'Regulating profiling in a democratic constitutional state' in Hildebrandt M and Gutwirth S (eds), Profiling the European Citizen (Springer 2008).

De Schutter O and Ringelheim J, 'Ethnic profiling: A rising challenge for European human rights law' (2008) 71(3) The Modern Law Review 358.

Dieterich W, Mendoza C and Brennan T, 'COMPAS risk scales: Demonstrating accuracy equity and predictive parity' (2016) Northpoint Inc.

Diaz C and Gürses S, 'Understanding the landscape of privacy technologies' (2012) www.cosic.esat.kuleuven.be/publications/article-2215.pdf accessed 10 October 2018.

Domingos P, The master algorithm: How the quest for the ultimate learning machine will remake our world (Basic Books 2015).

Dommering E, 'Regulating technology: code is not law', in Dommering E and Ascher L, Coding regulation: Essays on the normative role of information technology 1, Asser Press.

Dourish P, 'Algorithms and their others: Algorithmic culture in context' (2016) 3(2) Big Data & Society.

Dutch Association of Insurers, 'Grip op data: green paper Big Data' ['Understanding data: green paper Big Data']

https://www.verzekeraars.nl/media/1489/grip-op-data-green-paper-big-data.pdf accessed on 2 October 2018.

Dutch Data Protection Authority 2017, 'Dutch data protection authority: Facebook violates privacy law', 16 May 2017,

https://autoriteitpersoonsgegevens.nl/en/news/dutch-data-protection-authority-facebook-violates-privacy-law accessed 1 October 2018.

Dutch Data Protection Authority 2017a, 'Informal English translation of the conclusions of the Dutch Data Protection Authority in its final report of findings about its investigation into the processing of personal data by the Facebook group, 23 February 2017',

https://autoriteitpersoonsgegevens.nl/sites/default/files/atoms/files/conclusions facebook february 23 2017.pdf accessed 1 October 2018.

Duhigg C, 'How companies learn your secrets', New York Times, 16 February 2012<

https://www.nytimes.com/2012/02/19/magazine/shopping-habits.html accessed 10 October 2018.

Dwork C et al., 'Fairness through awareness' (Proceedings of the 3rd Innovations in Theoretical Computer Science Conference ACM, 2012) 214.

ECRI general policy recommendation no. 2 (2018): Equality Bodies to combat racism and intolerance at national level, 7 December 2017, Strasbourg, CRI (2018)06 https://rm.coe.int/ecri-general-policy-/16808b5a23 accessed 14 October 2018.

ECRI Statute Resolution 2002: Council of Europe Committee of Ministers, Resolution Res(2002)8 on the statute of the European Commission against Racism and Intolerance,

https://search.coe.int/cm/Pages/result_details.aspx?ObjectId=09000016805 e255a accessed 1 October 2018.

Edwards L and Veale M, 'Slave to the algorithm: Why a right to an explanation is probably not the remedy you are looking for' (2017) 16 Duke L.& Tech.Rev. 18.

Edwards L and Veale M, 'Enslaving the algorithm: from a "right to an explanation" to a "right to better decisions"?'. IEEE Security & Privacy, 2018, 16(3), 46-54.

Ellis E and Watson P, EU anti-discrimination law (OUP Oxford 2012).

European Agency for Fundamental Rights, Handbook on European data protection law (2018 edition) (Publications Office of the European Union 2018).

European Commission, Communication from the Commission to the European Parliament, the European Council, the Council, the European Economic and Social Committee and the Committee of the Regions, Brussels, 25.4.2018 COM(2018) 237 final, {Swd(2018) 137 Final} https://ec.europa.eu/digital-single-market/news-redirect/624220 accessed 26 September 2018.

European Consumer Organisation BEUC. 'Automated decision making and artificial intelligence - a consumer perspective, BEUC position paper, by Schmon C.' (June 2018) https://www.beuc.eu/publications/beuc-x-2018-058 automated decision making and artificial intelligence.pdf accessed on 1 October 2018.

European Data Protection Supervisor, 'Privacy and competitiveness in the age of big data: The interplay between data protection, competition law and consumer protection in the Digital Economy', March 2014,

https://edps.europa.eu/sites/edp/files/publication/14-03-

26 competitition law big data en.pdf accessed 26 September 2018.

European Data Protection Supervisor. 'EDPS Opinion on coherent enforcement of fundamental rights in the age of big data (Opinion 8/2016)' (23 September 2016)

https://secure.edps.europa.eu/EDPSWEB/webdav/site/mySite/shared/Documents/EDPS/Events/16-09-23 BigData opinion EN.pdf accessed 24 September 2018.

Equivant, COMPAS Classification, 2018, http://www.equivant.com/solutions/inmate-classification accessed 19 September 2018.

European Group on Ethics in Science and New Technologies, 'Statement on Artificial Intelligence, Robotics and 'Autonomous' Systems', March 2018, https://ec.europa.eu/research/ege/pdf/ege ai statement 2018.pdf accessed 26 September 2018.

Ezrachi A and Stucke ME, Virtual Competition (Harvard 2016)

Federal Trade Commission. 'Big data: A tool for inclusion or exclusion? Understanding the issues' (January 2016)

https://www.ftc.gov/system/files/documents/reports/big-data-tool-inclusion-or-exclusion-understanding-issues/160106big-data-rpt.pdf accessed on 14 October 2018.

Feller A et al., 'A computer program used for bail and sentencing decisions was labeled biased against blacks. It's actually not that clear' (2016) The Washington Post.

Ferguson AG, The Rise of Big Data Policing: Surveillance, Race, and the Future of Law Enforcement (NYU Press 2017).

Ferraris V et al., 'Defining Profiling' (Profiling Project working paper 2013) http://dx.doi.org/10.2139/ssrn.2366564 accessed 3 October 2018.

Frawley WJ, Piatetsky-Shapiro G and Matheus CJ, 'Knowledge discovery in databases: An overview' (1992) 13(3) Al magazine 57.

Fiesler C, 'Tech ethics curricula: A collection of syllabi', https://medium.com/@cfiesler/tech-ethics-curricula-a-collection-of-syllabi-3eedfb76be18 accessed 29 September 2018.

Financial Conduct Authority. 'Feedback statement: call for inputs on Big Data in retail general insurance' (September 2016) www.fca.org.uk/sites/default/files/fs16-05-big-data-retail-general-insurance.pdf accessed on 23 December 2016.

Fink K, 'Opening the government's black boxes: freedom of information and algorithmic accountability' (2018) 21(10) Information, Communication & Society 1453.

Fortune, 'Here Are the Fortune 500's 10 Most Valuable Companies', http://fortune.com/2018/05/21/fortune-500-most-valuable-companies-2018/ accesses 24 September 2018.

Fredman S. 'Intersectional discrimination in EU gender equality and nondiscrimination law, European network of legal experts in gender equality and non-discrimination' (Report for European Commission, Directorate-General for Justice and Consumers), May 2016

https://publications.europa.eu/en/publication-detail/-/publication/d73a9221-b7c3-40f6-8414-8a48a2157a2f/language-en accessed on 9 October 2018.

Friedman B and Nissenbaum H, 'Bias in computer systems' (1996) 14(3) ACM Transactions on Information Systems (TOIS) 330.

Frucci A, 'HP Face-Tracking Webcams Don't Recognize Black People', Gizmodo 21 December 2009, https://gizmodo.com/5431190/hp-face-tracking-webcams-dont-recognize-black-people accessed 1 October 2018.

Gama J et al., 'A Survey on concept drift adaptation' (2014) 46 ACM Comput. Surv. 44.

Gandy OH, The Panoptic Sort: A Political Economy of Personal Information (Westview 1993).

Gellert, R, De Vries, K, De Hert, P, and Gutwirth, S, 'A comparative analysis of anti-discrimination and data protection legislations', in Discrimination and privacy in the information society (pp. 61-89). Springer, Berlin, Heidelberg 2013.

Gerards JH, 'Discrimination grounds', in M. Bell and D. Schiek (eds.), lus commune case books for a common law of Europe – Non-discrimination (Hart 2007), p. 33-184.

Goodman B, 'Discrimination, data sanitisation and auditing in the European Union's General Data Protection Regulation' (2016) 2 Eur.Data Prot.L.Rev. 493.

Goodman B and Flaxman S, 'European Union regulations on algorithmic decision-making and a "right to explanation" (2016) arXiv preprint arXiv:1606.08813.

Graef I, EU Competition Law, Data Protection and online platforms: Data as essential facility (Kluwer Law International 2016).

Graef I, 'Algorithms and fairness: What role for competition law in targeting price discrimination towards end consumers?' (2017)

https://ssrn.com/abstract=3090360 accessed 26 September 2018.

Greene D, Hoffman AL, Stark L, 'Better, nicer, clearer, fairer: A critical assessment of the movement for ethical artificial intelligence and machine learning', http://dmgreene.net/wp-content/uploads/2018/09/Greene-Hoffman-Stark-Better-Nicer-Clearer-Fairer-HICSS-Final-Submission.pdf 2018, accessed 26 September 2018.

Greenleaf G, 'Global tables of data privacy laws and bills (5th Ed 2017)'. 145 Privacy Laws & Business International Report, 14-26, https://ssrn.com/abstract=2992986 accessed 8 September 2018.

Guidotti R and others, 'A survey of methods for explaining black box models' (2018) 51(5) ACM Computing Surveys (CSUR) 93.

Gürses S and Van Hoboken J, 'Privacy after the agile turn' (2017) Cambridge Handbook of Consumer Privacy. Cambridge University Press, Cambridge, https://osf.io/ufdvb/ accessed 1 October 2018.

Hacker P, 'Teaching fairness to artificial intelligence: Existing and novel strategies against algorithmic discrimination under EU law' (2018) 55 Common Market Law Review, Issue 4, pp. 1143–1185.

Han J, Pei J and Kamber M, Data mining: concepts and techniques (Elsevier 2011).

Harcourt BE, Against prediction: Profiling, policing, and punishing in an actuarial age (University of Chicago Press 2008).

Hardt M, 'How big data is unfair. Understanding sources of unfairness in data driven decision making' (2014) https://medium.com/@mrtz/how-big-data-is-unfair-9aa544d739de accessed on 2 October 2018.

Helberger N., Zuiderveen Borgesius F.J. and Reyna A., 'The perfect match? A closer look at the relationship between EU consumer law and data protection law', Common Market Law Review, 2017-54, p. 1427-1466.

Hildebrandt M, 'Defining profiling: a new type of knowledge?' in Hildebrandt M and Gutwirth S (eds), Profiling the European citizen (Springer 2008).

Hildebrandt M, Criminal Law and Technology in a Data-Driven Society, in: M Dubber and T Hörnle, The Oxford Handbook of Criminal Law, Oxford University Press 2014, p. 174-197.

Hildebrandt, M, Smart technologies and the end (s) of law: Novel entanglements of law and technology, Edward Elgar Publishing 2015.

Hildebrandt M and Gutwirth S (eds), Profiling the European Citizen (Springer 2008).

Hirsch DD, 'The law and policy of online privacy: Regulation, self-regulation, or co-regulation' (2010) 34 Seattle UL Rev. 439.

Jabłonowska A et al., 'Consumer law and artificial intelligence: challenges to the EU consumer law and policy stemming from the business' use of artificial intelligence: final report of the ARTSY project' (2018) http://cadmus.eui.eu/bitstream/handle/1814/57484/WP_2018_01.pdf?seque

nce=1&isAllowed=y accessed 26 September 2018.

Jernigan C and Mistree BF, 'Gaydar: Facebook friendships expose sexual orientation' (2009) 14(10) First Monday.

Jordan M, 'Artificial intelligence-The revolution hasn't happened yet' (April 2018) https://medium.com/@mijordan3/artificial-intelligence-the-revolution-hasnt-happened-yet-5e1d5812e1e7 accessed 1 August 2018.

Kay M, Matuszek C and Munson SA, 'Unequal representation and gender stereotypes in image search results for occupations' (Proceedings of the 33rd Annual ACM Conference on Human Factors in Computing Systems ACM, 2015) 3819.

Kaminski M 2018, 'The right to explanation, explained' (2018). https://osf.io/preprints/lawarxiv/rgeus/ accessed 2 October 2018.

Kaminski M 2018a, 'Binary governance: A two-part approach to accountable algorithms' (2018). 92 S. Calif. L. Rev (forthcoming 2019).

Kamiran F and Calders T, 'Classifying without discriminating', 2009 2nd International Conference on Computer, Control and Communication. (17-18, February 2009): 1–6. doi:10.1109/IC4.2009.4909197

Kamiran F and Calders T, 'Data preprocessing techniques for classification without discrimination' (2012) 33(1) Knowledge and Information Systems 1.

Khaitan T, A theory of discrimination law (OUP Oxford 2015).

Kilbertus N. et al., 'Blind justice: Fairness with encrypted sensitive attributes' (2018) Proceedings of the 35th International Conference on Machine Learning (ICML 2018).

Kim PT, 'Data-driven discrimination at work' (2016) 58 Wm.& Mary L.Rev. 857.

Kloza D, Van Dijk N, Gellert R, I Maurice, Borocz I, Tanas A, Mantovani E, Quinn P, Data protection impact assessments in the European Union: complementing the new legal framework towards a more robust protection of individuals. d.pia.lab Policy Brief Article 2017, pp. 1-4. https://cris.vub.be/files/32009890/dpialab_pb2017_1_final.pdf accessed 25 September 2018.

Koene A et al., 'IEEE P70xx, Establishing standards for ethical technology' (2018) Lu Wang (East China Normal University).

Koops BJ, 'Should ICT regulation be technology-neutral?' in Koops, Bert-Jaap et al. (eds), Starting Points for ICT Regulation - Deconstructing Prevalent Policy One-liners (Information Technology and Law Series, Asser 2006). https://ssrn.com/abstract=918746 accessed 10 October 2018.

Koops BJ, 'Some reflections on profiling, power shifts, and protection paradigms' in Hildebrandt M and Gutwirth S (eds), Profiling the European citizen: cross-disciplinary perspectives (Springer 2008).

Korff D, 'Comments on Selected Topics in the Draft EU Data Protection Regulation' (17 September 2012) http://ssrn.com/abstract=2150145 accessed 25 September 2018.

Kroll JA et al., 'Accountable algorithms' (2016) 165 University of Pennsylvania Law Review 633-705.

Lammerant H, de Hert P and Blok P, 'Big data en gelijke behandeling (Big data and equal treatment)' in Blok, P (ed), Big data & het recht (Sdu Uitgeverij) 2017.

https://cris.vub.be/files/35864071/pdh17 hlpb big data en gelijke behand eling H6 Blok.pdf accessed 15 October 2018.

Lippert-Rasmussen K, Born free and equal? A philosophical inquiry into the nature of discrimination (Oxford University Press 2014).

Kusner MJ et al., 'Counterfactual fairness' (Advances in Neural Information Processing Systems 2017) 4066.

Larson J et al., 'These are the job ads you can't see on Facebook if you're older', The New York Times, 19 December 2017.

Larson J et al., 'How we analyzed the COMPAS recidivism algorithm' ProPublica, May 2016 https://www.ProPublica.org/article/how-we-analyzed-the-compas-recidivism-algorithm accessed 1 August 2018.

Larson J, Mattu S and Angwin J, 'Unintended consequences of geographic targeting', Technology Science 2015.

Laskov P and Lippmann R, 'Machine learning in adversarial environments' (2010) 81 Machine Learning 115

Lehr D and Ohm P, 'Playing with the data: What legal scholars should learn about machine learning' (2017) 51 UCDL Rev. 653, p, 671.

Lipton ZC, 'From AI to ML to AI: On swirling nomenclature & slurried thought' (5 June 2018) http://approximatelycorrect.com/2018/06/05/ai-ml-ai-swirling-nomenclature-slurried-thought/ accessed 1 August 2018.

Lowry S and Macpherson G, 'A blot on the profession' (1988) 296(6623) Br Med J (Clin Res Ed) 657.

Lum K and Isaac W, 'To predict and serve?' (2016) 13(5) Significance 14.

Lyon D, 'Surveillance as social sorting: computer codes and mobile bodies 'in Lyon, D. (ed), Surveillance as social sorting: privacy, risk and automated discrimination (Routledge 2002).

Malgieri G, 'Trade secrets v personal data: a possible solution for balancing rights' (2016) 6 International Data Privacy Law 2, 103.

Malgieri G, 'Right to explanation and algorithm legibility in the EU Member States legislations', 17 August 2018, https://ssrn.com/abstract=3233611 accessed 10 October 2018.

Malgieri G and Comandé G 2017, 'Why a right to legibility of automated decision-making exists in the general data protection regulation' (2017) International Data Privacy Law.

Malgieri G and Comandé G 2017a, 'Sensitive-by-distance: quasi-health data in the algorithmic era' (2017) 26(3) Information & Communications Technology Law 229.

Mantelero A, 'Artificial Intelligence and data protection: Challenges and possible remedies', draft report for the Council of Europe's Consultative Committee of the Convention for the Protection of Individuals with regard to Automatic Processing of Personal Data, 17 September 2018, T-PD(2018)09Rev https://rm.coe.int/report-on-artificial-intelligence-artificial-intelligence-and-data-pro/16808d78c9 accessed 30 September 2018.

Mantelero A, 'Al and Big Data: A blueprint for a human rights, social and ethical impact assessment' (2018) 34(4) Computer Law & Security Review 754.

Mattioli, D, 'On Orbitz, Mac users steered to pricier hotels', Wall Street Journal, 23 August 2012.

McCarthy J et al., 'A proposal for the Dartmouth summer research project on artificial intelligence, august 31, 1955, http://www-

formal.stanford.edu/jmc/history/dartmouth/dartmouth.html accessed 1 October 2018.

McCray LE, Oye KA and Petersen AC, 'Planned adaptation in risk regulation: An initial survey of US environmental, health, and safety regulation' (2010) 77 Technological Forecasting and Social Change 951

Mendoza I and Bygrave LA, 'The right not to be subject to automated decisions based on profiling' (2017).

Miller T, 'Explanation in artificial intelligence: insights from the social sciences' (2017) arXiv preprint arXiv:1706.07269.

Mittelstadt BD et al., 'The ethics of algorithms: Mapping the debate' (2016) 3(2) Big Data & Society 2053951716679679.

Munoz C, Smith M and Patil DJ, 'Big data: A report on algorithmic systems, opportunity, and civil rights' (White House, Executive Office of the President) 2016

https://obamawhitehouse.archives.gov/sites/default/files/microsites/ostp/201 6 0504 data discrimination.pdf accessed 1 October 2018.

Narayanan A, 'Language necessarily contains human biases, and so will machines trained on language corpora', Freedom to Tinker 24 August 2016, https://freedom-to-tinker.com/2016/08/24/language-necessarily-contains-human-biases-and-so-will-machines-trained-on-language-corpora/ accessed 29 September 2018.

Naudts, L, 'Fair or unfair algorithmic differentiation? Luck Egalitarianism As a Lens for Evaluating Algorithmic Decision-Making', 18 August 2017, https://ssrn.com/abstract=3043707 accessed 2 October 2018.

Nemitz, P, 'Constitutional democracy and technology in the age of artificial intelligence', Phil. Trans. R. Soc. A 376.2133 (2018): 20180089, http://rsta.royalsocietypublishing.org/content/376/2133/20180089 accessed 16 October 2018.

Noble SU, Algorithms of Oppression: How search engines reinforce racism (NYU Press 2018).

O'Neil C, Weapons of math destruction: How big data increases inequality and threatens democracy (Crown Publishing Group (NY) 2016).

Oswald M, 'Algorithm-assisted decision-making in the public sector: framing the Issues using administrative law rules governing discretionary power' (2018) 376 Philosophical Transactions of the Royal Society A: Mathematical, Physical and Engineering Sciences 20170359.

Oswald M and Grace J, 'Intelligence, Policing and the Use of Algorithmic Analysis: A Freedom of Information-Based Study' (2016) 1 Journal of Information Rights, Policy and Practice.

Parasuraman R and Manzey DH, 'Complacency and bias in human use of automation: An attentional integration' (2010) 52(3) Hum Factors 381.

Pasquale F, The black box society: The secret algorithms that control money and information (Harvard University Press 2015).

Pasquale F, 'Toward a fourth law of robotics: Preserving attribution, responsibility, and explainability in an algorithmic society', 14 July 2017, Ohio State Law Journal, Vol. 78, 2017; https://ssrn.com/abstract=3002546 accessed 1 October 2018.

Paul A, Jolley C, and Anthony A. 'Reflecting the past, shaping the future: Making AI work for international development (report United States Agency for International Development)' (2018)

https://www.usaid.gov/sites/default/files/documents/15396/AI-ML-in-Development.pdf accessed 1 October 2018.

Peck D, 'They're watching you at work', The Atlantic, December 2013, https://www.theatlantic.com/magazine/archive/2013/12/theyre-watching-you-at-work/354681/ accessed 1 October 2018.

Pedreschi D, Ruggieri S, Turini F, 'Discrimination-aware data mining' (2008) ACM Int. Conf. on Knowledge Discovery and Data Mining (KDD 2008) 560–568:

Peña Gangadharan S and Niklas J, 'Between antidiscrimination and data: understanding human rights discourse on automated discrimination in Europe' (2018).

Peppet SR, 'Regulating the Internet of Things: first steps toward managing discrimination, privacy, security, and consent' (2014) 93 Texas Law Review 85.

Perry WL et al., 'Predictive policing: The role of crime forecasting in law enforcement operations' (2013)

https://www.rand.org/content/dam/rand/pubs/research_reports/RR200/RR23 3/RAND_RR233.pdf accessed on 29 September 2017.

Poole DL, Mackworth AK and Goebel R, Computational intelligence: a logical approach (Oxford University Press New York 1998).

Prates M, Avelar P and Lamb L, 'Assessing gender bias in machine translation - A case study with Google translate' 2018 https://arxiv.org/abs/1809.02208 accessed 11 September 2018, p. 1.

Puppe F, Systematic introduction to expert systems: Knowledge representations and problem-solving methods (Springer Science & Business Media 1993).

Raab C and Szekely I, 'Data protection authorities and information technology' (2017) 33(4) Computer Law & Security Review 421.

Regan J, 'New Zealand passport robot tells applicant of Asian descent to open eyes', Reuters 7 December 2016, https://www.reuters.com/article/us-china-tencent/tencent-announces-a-restructuring-as-challenges-rise-idUSKCN1MA04T accessed 1 October 2018.

Reisman D, Schultz J, Crawford K, and Whittaker M, 'Algorithmic impact assessments: A practical framework for public agency accountability' (Al Now Institute 2018).

Rieke A, Bogen M and Robinson DG, 'Public scrutiny of automated decisions: Early lessons and emerging methods, Upturn and Omidyar Network' (2018)

http://www.omidyar.com/sites/default/files/file_archive/Public%20Scrutiny%2 0of%20Automated%20Decisions.pdf accessed on 3 October 2018.

Rieke A, Robinson DG and Yu H, 'Civil rights, big data, and our algorithmic future: A September 2014 report on social justice and technology (Version 1.2), Washington, DC: Upturn, PDF version, https://bigdata.fairness.io accessed 1 October 2018.

Ringelheim J and De Schutter O, 'The processing of racial and ethnic data in antidiscrimination policies: Reconciling the promotion of equality with privacy rights' (2009) Brussels, Bruylant.

Robinson, D. and L. Koepke. 'Stuck in a pattern' (2016)

https://www.upturn.org/static/reports/2016/stuck-in-a-pattern/files/Upturn_-

Stuck In a Pattern v.1.01.pdf accessed on 4 October 2018.

Royal Society (UK). 'Machine learning: the power and promise of computers that learn by example' (April 2017)

https://royalsociety.org/~/media/policy/projects/machine-

learning/publications/machine-learning-report.pdf accessed 1 May 2017.

Russell SJ and Norvig P, Artificial intelligence: a modern approach (third edition) (Prentice Hall 2016).

Sandvig C et al., 'Auditing algorithms: Research methods for detecting discrimination on internet platforms' (2014) Data and discrimination: converting critical concerns into productive inquiry 1.

Selbst AD, 'Disparate impact in Big Data policing' (2017) 52 Ga.L.Rev. 109.

Selbst AD and Powles J, 'Meaningful information and the right to explanation' (2017) 7(4) International Data Privacy Law 233.

Selbst AD and Barocas S, 'The intuitive appeal of explainable machines', 2018, Fordham Law Review, Forthcoming.

https://ssrn.com/abstract=3126971 accessed 11 October 2018.

Sharp G, 'Nikon camera says asians: people are always blinking', https://thesocietypages.org/socimages/2009/05/29/nikon-camera-says-asians-are-always-blinking/ 29 May 2009, accessed 1 October 2018.

Schauer FF, Profiles, probabilities, and stereotypes (Harvard University Press 2003).

Schreurs W, Hildebrandt M, Kindt, E, and Vanfleteren M, 'Cogitas, ergo sum. The role of data protection law and non-discrimination law in group profiling in the private sector' in Profiling the European citizen (Springer 2008).

Siegel E, Predictive analytics: The power to predict who will click, buy, lie, or die (John Wiley & Sons 2013).

Sunstein CR, 'Problems with rules' (1995) 83(4) California Law Review 953.

Swedloff R, 'Risk classification's Big Data (r)evolution' (2014) 21 Connecticut Insurance Law Journal 339.

Sweeney L, 'Discrimination in online ad delivery' (2013) 11(3) ACM Queue 10.

Taylor L, 'What is data justice? The case for connecting digital rights and freedoms globally' (2017) 4(2) Big Data & Society

Taylor L, van der Sloot B, Floridi L (eds.), Group privacy: New challenges of data technologies (Springer 2017).

Tene O and Polonetsky J, 'Taming the Golem: Challenges of ethical algorithmic decision-making' (2017) 19 NCJL & Tech. 125.

Tickle AB et al., 'The truth will come to light: Directions and challenges in extracting the knowledge embedded within trained artificial neural networks' (1998) 9(6) IEEE Trans Neural Networks 1057.bler

Tobler C, Indirect discrimination: a case study into the development of the legal concept of indirect discrimination under EC law, vol 10 (Intersentia 2005).

Turing A, 'Can digital computers think?', 1951, reprinted in Copeland, BJ, The essential Turing (Clarendon Press 2004).

Turow J, The Daily You: How the new advertising industry is defining your identity and your worth (Yale University Press 2011).

Valcke P, Graef I and Clifford D, 'iFairness – Constructing fairness in IT (and other areas of) law through intra-and interdisciplinarity' (2018) 34(4) Computer Law & Security Review 707.

Valentino-Devries, J., Singer-Vine, J., and Soltani, A, 'Websites vary prices, deals based on users' information', Wall Street Journal, 23 December 2012.

Van Brakel R and De Hert P, 'Policing, surveillance and law in a pre-crime society: Understanding the consequences of technology based strategies.' (2011) 20 Technology-led policing 165.

Van Eck, B.M.A., Geautomatiseerde ketenbesluiten and rechtsbescherming: Een onderzoek naar de praktijk van geautomatiseerde ketenbesluiten over een financieel belang in relatie tot rechtsbescherming [Automated administrative chain decisions and legal protection. Research into legal safeguards regarding the practice of automated chain decisions about financial interests] PhD thesis University of Tilburg, https://pure.uvt.nl/portal/files/20399771/Van_Eck_Geautomatiseerde_ketenb

https://pure.uvt.nl/portal/files/20399771/Van_Eck_Geautomatiseerde_ketenbesluiten.pdf accessed 26 September 2018.

Van Nooren P, Van Gorp N, Van Eijk N, Fathaigh R, 'Should we regulate digital platforms? A new framework for evaluating policy options', Policy & Internet 2018, 264-301,

https://www.ivir.nl/publicaties/download/Policy and Internet 2018.pdf accessed 15 October 2018.

Veale M and Binns R., 'Fairer machine learning in the real world: Mitigating discrimination without collecting sensitive data' (2017) 4(2) Big Data & Society.

Veale M, Binns R., and Edwards L. 'Algorithms that remember: Model inversion attacks and data protection law' (2018) 376 Philosophical Transactions of the Royal Society A: Mathematical, Physical and Engineering Sciences 20180083.

Veale M, Van Kleek M and Binns R, 'Fairness and accountability design needs for algorithmic support in high-stakes public sector decision-making' (2018) Proceedings of the 36rd Annual ACM Conference on Human Factors in Computing Systems (CHI 2018).

Veale M and Edwards L, 'Clarity, surprises, and further questions in the Article 29 Working Party Draft Guidance on automated decision-making and profiling' (2018) 34 Computer Law & Security Review 398.

Vedder, AH, 'Privatization, information technology and privacy: Reconsidering the social responsibilities of private organizations', in Moore G (ed.), Business ethics: Principles and practice, 215–226 (Sunderland: Business Education Publishers 1997).

Vetzo M, Gerards J, and Nehmelman R. Algoritmes en grondrechten [Algorithms and fundamental rights], Utrecht University/Boom Juridisch (2018) https://www.uu.nl/sites/default/files/rebo-montaigne-algoritmes en grondrechten.pdf accessed on 12 October 2018.

Wachter S, 'Normative challenges of identification in the Internet of Things: Privacy, profiling, discrimination, and the GDPR', Computer Law & Security Review, Volume 34, Issue 3, June 2018, p. 436-449.

Wachter S and Mittelstadt B, 'A right to reasonable inferences: Re-thinking data protection law in the age of Big Data and Al', September 13, 2018, Columbia Business Law Review, forthcoming, https://ssrn.com/abstract=3248829 accessed 9 October 2018.

Wachter S, Mittelstadt B and Russell C, 'Counterfactual explanations without opening the black box: automated decisions and the GDPR' (2017) 31(2 Spring 2018) Harvard Journal of Law & Technology 841.

Wachter S, Mittelstadt, B, and Floridi, L, 'Why a right to explanation of automated decision-making does not exist in the general data protection regulation', International Data Privacy Law 2017-2.

Wagner B (2018). Ethics as an Escape from Regulation: From ethics-washing to ethics-shopping? In M. Hildebrandt (Ed.), Being Profiling. Cogitas ergo sum. Amsterdam University Press, draft available at https://www.privacylab.at/wp-content/uploads/2018/07/Ben Wagner Ethics-as-an-Escape-from-Regulation 2018 BW9.pdf accessed 24 September 2018.

Wagner B. et al. 'Algorithms and human rights. Study on the human rights dimensions of automated data processing techniques and possible regulatory implications, DGI(2017)12, prepared by the Committee of Experts on internet intermediaries (MSI-NET) for the Council of Europe' (2018) https://rm.coe.int/algorithms-and-human-rights-en-rev/16807956b5 accessed 12 June 2018.

Wakefield A and Fleming J, 'Responsibilization', The SAGE dictionary of policing (SAGE Publications Ltd 2009).

Wright D, De Hert P (eds.) Privacy Impact Assessment (Springer 2012).

York C, 'Three black teenagers: Is Google Racist? It's not them, it's us', 8 June 2016, https://www.huffingtonpost.co.uk/entry/three-black-teenagers-google-racism_uk_575811f5e4b014b4f2530bb5 accessed 1 October 2018.

Zarsky TZ, 'Mine your own business: making the case for the implications of the data mining of personal information in the forum of public opinion' (2002) 5 Yale Journal of Law and Technology 1.

Zarsky TZ, 'An analytic challenge: discrimination theory in the age of predictive analytics' (2017) 14 ISJLP 11.

Zarsky TZ, 'The trouble with algorithmic decisions - An analytic road map to examine efficiency and fairness in automated and opaque decision making' (2015) Science, Technology & Human Values 0162243915605575.

Žliobaitė I and Custers B, 'Using sensitive personal data may be necessary for avoiding discrimination in data-driven decision models' (2016) 24(2) Artificial Intelligence and Law 183.

Zuiderveen Borgesius 2015, 'Behavioural sciences and the regulation of privacy on the Internet', in A-L Sibony and A. Alemanno (eds.), Nudge and the law - what can EU law learn from behavioural sciences? (Hart Publishing 2015), p. 179-207. https://ssrn.com/abstract=2513771 accessed 25 September 2018.

Zuiderveen Borgesius 2015a, Improving privacy Protection in the Area of Behavioural Targeting, Kluwer law International.

http://hdl.handle.net/11245/1.434236 accessed 25 September 2018.

Zuiderveen Borgesius FJ and Poort J, 'Online price discrimination and EU data privacy law', Journal of Consumer Policy, 2017, p. 1-20. https://ssrn.com/abstract=3009188 accessed 25 September 2018.

Zuiderveen Borgesius FJ, Gray J and van Eechoud M, 'Open data, privacy, and Fair Information Principles: Towards a balancing framework' (2015) 30 Berkeley Tech.LJ 2073. https://ssrn.com/abstract=2695005 accessed 25 September 2018.

Zuiderveen Borgesius FJ, Van Hoboken J, Fahy R, Irion K, Rozendaal M, 'An assessment of the Commission's proposal on privacy and electronic communications', Directorate-General for Internal Policies, Policy Department C: Citizen's Rights and Constitutional Affairs, May 2017. https://ssrn.com/abstract=2982290 accessed 26 September 2018.

Directorate General of Democracy

Council of Europe Tel.: +33 (0) 3 90 21 46 62

www.coe.int

The Council of Europe is the continent's leading human rights organisation. It comprises 47 member states, including all members of the European Union. All Council of Europe member states have signed up to the European Convention on Human Rights, a treaty designed to protect human rights, democracy and the rule of law. The European Court of Human Rights oversees the implementation of the Convention in the member states.

> COUNCIL OF EUROPE CONSEIL DE L'EUROPE